UNITED STATES SECURITIES AND EXCHANGE COMMISSION

WASHINGTON, D.C. 20549

SCHEDULE 13D

(Rule 13d-101)

INFORMATION TO BE INCLUDED IN STATEMENTS FILED PURSUANT TO § 240.13d-1(a) AND AMENDMENTS THERETO FILED PURSUANT TO §240.13d-2(a)

UNDER THE SECURITIES ACT OF 1934 (Amendment No.)*

COLFAX CORPORATION

(Name of Issuer)

Common Stock, par value \$0.001 per share (Title of Class of Securities)

194014106 (CUSIP Number)

William R. Bush
BDT Capital Partners, LLC
401 N. Michigan Ave., Suite 3100
Chicago, Illinois
(312) 660-7300
(Name, Address and Telephone Number of Person Authorized to Receive Notices and Communications)

January 13, 2012

If the filing person has previously filed a statement on Schedule 13G to report the acquisition that is the subject of this Schedule 13D, and is filing this schedule because of §§ 240.13d-1(e), 240.13d-1(f) or 240.13d-1(g), check the following box. \Box

(Date of Event Which Requires Filing of this Statement)

Note. Schedules filed in paper format shall include a signed original and five copies of the schedule, including all exhibits. See § 240.13d-7 for other parties to whom copies are to be sent.

* The remainder of this cover page shall be filled out for a reporting person's initial filing on this form with respect to the subject class of securities, and for any subsequent amendment containing information which would alter disclosures provided in a prior cover page.

The information required on the remainder of this cover page shall not be deemed to be "filed" for the purpose of Section 18 of the Securities Exchange Act of 1934 ("Act") or otherwise subject to the liabilities of that section of the Act but shall be subject to all other provisions of the Act (however, see the Notes).

CUSIP	USIP No. 194014106 Page 2 of 6 Pages				
(1)	Name of reporting persons				
	BDT Capital Partners, LLC				
(2)	Check the appropriate box if a member of a group* (a) ⊠ (b) □				
(3)	SEC use	only			
(4)	Source o	of fund	ls*		
	00, W0	2			
(5)	Check E	Sox if o	disclosure of legal proceedings is required pursuant to Items 2(d) or 2(e)		
(6)	Citizens	hip or	place of organization		
	Delawai	e			
(7) Sole voting power					
Nu	mber of		0 shares		
	hares eficially	(8)	Shared voting power		
	ned by		26,930,236 shares		
	each	(9)	Sole dispositive power		
reporting person			0 shares		
-		(10)	Shared dispositive power		
	26,930,236 shares				
(11)					
	26,930,236 shares				
(12)	2) Check box if the aggregate amount in Row (11) excludes certain shares* □				
(13)	Percent	of clas	s represented by amount in Row (11)		
	27.8%1				
(14)	Type of reporting person*				

^{*} See Instructions

Percentage of beneficial ownership is based upon the number of shares of the issuer's common stock that will be outstanding assuming the closing of certain equity issuances by the issuer related to its acquisition of Charter International plc described in Item 4 of this Schedule 13D.

CUSIF	USIP No. 194014106 Page 3 of 6 Pages				
(1)	Name of reporting persons				
	BDTCP GP I, LLC				
(2)			propriate box if a member of a group*		
()	(a) ⊠	(b)			
(3)	SEC use	only			
(3)		J			
(4)	Source o	of fund	ds*		
	00, W0	2			
(5)			disclosure of legal proceedings is required pursuant to Items 2(d) or 2(e) \Box		
(C)	C:+:	1-1			
(6)	Citizens	inb or	place of organization		
	Delawar	e			
(7) Sole voting power					
Nu	mber of		0 shares		
	hares	(8)	Shared voting power		
	eficially ned by		26,930,236 shares		
	each	(9)			
	porting				
	erson with	(10)	0 shares Shared dispositive power		
with (10) Shared dispositive power					
	26,930,236 shares				
(11)	Aggregate amount beneficially owned by each reporting person				
	26,930,236 shares				
(12)	2) Check Box if the aggregate amount in Row (11) excludes certain shares* □				
(13)	3) Percent of class represented by amount in Row (11)				
(10)	5) Terecite of class represented by annount in Now (11)				
	27.8% ²				
(14)	Type of	reporti	ing person*		
	00				

^{*} See Instructions
² See footnote 1 al See footnote 1 above.

CUSIF	USIP No. 194014106 Page 4 of 6 Pages					
(1)	Name of reporting persons					
	BDT CF Acquisition Vehicle, LLC					
(2)						
()	(a) ⊠	(b)				
(3)	SEC use	only				
(3)		J				
(4)	Source o	of fund	ds*			
	00, W0	2				
(5)			disclosure of legal proceedings is required pursuant to Items 2(d) or 2(e)			
(E)	Citizana	hin or	place of organization			
(6)	Citizens	шр ог	place of organization			
	Delawar					
(7) Sole voting power						
Nu	mber of		26,930,236 shares			
	hares	(8)	Shared voting power			
	eficially ned by		0 shares			
	each	(9)				
_	porting		26.020.226 - 1			
	erson with	(10)	26,930,236 shares Shared dispositive power			
(10) Shared dispositive power		(10)	Shared dispositive power			
	0 shares					
(11)	Aggregate amount beneficially owned by each reporting person					
	26,930,236 shares					
(12)	2) Check Box if the aggregate amount in Row (11) excludes certain shares* □					
(13)	3) Percent of class represented by amount in Row (11)					
(-)						
(1.4)	27.8%3					
(14)	Type of reporting person*					
	00					

^{*} See Instructions
³ See footnote 1 al See footnote 1 above.

CUSIF	USIP No. 194014106 Page 5 of 6 Pages				
(1)	Name of reporting persons				
	Byron D. Trott				
(2)	ū		ropriate box if a member of a group*		
(-)	(a) □	(b)			
(2)	SEC use	only			
(3)	SEC use	Olliy			
(4)	Source o	of fund	ls*		
	AF				
(5)		ox if o	disclosure of legal proceedings is required pursuant to Items 2(d) or 2(e)		
(3)					
(6)	Citizens	hip or	place of organization		
	United S	States o	of America		
		(7)			
	mber of hares	(8)	0 shares Shared voting power		
	eficially	(0)	Shared voling power		
ow	ned by		26,930,236 shares		
	each	(9)	Sole dispositive power		
_	porting erson		0 shares		
	with	(10)			
(11)	26,930,236 shares 1) Aggregate amount beneficially owned by each reporting person				
(11)	Aggregate amount beneficianly owned by each reporting person				
	26,930,236 shares				
(12)	2) Check Box if the aggregate amount in Row (11) excludes certain shares* □				
(13)	3) Percent of class represented by amount in Row (11)				
()	5) Telection of class represented by amount in 10011 (11)				
(4.4)	27.8%4				
(14)	Type of	reporti	ing person*		
	IN				

^{*} See Instructions

4 See footnote 1 al

See footnote 1 above.

CUSIP	JSIP No. 194014106 Page 6 of 6 Pages				
(1)	Name of reporting persons				
(2)	BDTP GP, LLC Check the appropriate box if a member of a group*				
(2)	(a)) ⊠		
	` ′	,			
(3)	SEC use	only			
(4)	Source o	of fund	ds*		
` /					
/=\	AF				
(5)	Check E	Sox if (disclosure of legal proceedings is required pursuant to Items 2(d) or 2(e) \Box		
(6)	Citizens	hip or	place of organization		
	Dolor				
	Delawar	(7)	Sole voting power		
(7) Solie voting power					
Nu	mber of		0 shares		
	hares	(8)	Shared voting power		
	eficially ned by		26,930,236 shares		
	each	(9)	· · ·		
reporting Solid dispositive power					
-	erson		0 shares		
	with	(10)	Shared dispositive power		
	26,930,236 shares				
(11)	Aggregate amount beneficially owned by each reporting person				
	26,930,236 shares				
(12)					
(1=)	-) Sheet 2011 are approprie univarie in 1011 (11) excludes certain shares				
(13)	3) Percent of class represented by amount in Row (11)				
	27.8%5				
(14)		reporti	ting person*		
	00				

^{*} See Instructions

⁵ See footnote 1 above.

Item 1. Security and Issuer.

This statement relates to the common stock, par value \$0.001 per share (the "<u>Common Stock</u>"), of Colfax Corporation, a Delaware corporation (the "<u>Company</u>"). The principal executive offices of the Company are located at 8170 Maple Lawn Blvd., Suite 180, Fulton, Maryland, 20759.

Item 2. Identity and Background.

- (a) This statement is being filed jointly by (i) BDT Capital Partners, LLC, a Delaware limited liability company ("BDT CP"), (ii) BDTCP GP I, LLC, a Delaware limited liability company ("BDTCP GP I"), (iii) BDT CF Acquisition Vehicle, LLC, a Delaware limited liability company (the "Investor"), (iv) Byron D. Trott, and (v) BDTP GP, LLC, a Delaware limited liability company ("BDTP" and, together with Mr. Trott, BDT CP, BDTCP GP I and the Investor, the "Reporting Persons"). Byron D. Trott is the sole member of BDTP, which is the managing member of BDT CP. BDT CP is the manager of BDTCP GP I, which is the manager of the Investor.
- **(b)** The address of the principal business and principal office of each of the Reporting Persons is 401 North Michigan Avenue, Suite 3100, Chicago, Illinois 60611.
- (c) The principal occupation of Mr. Trott is President and Chief Investment Officer of BDT CP. The principal business of each of the other Reporting Persons is investing.
- **(d)** None of the Reporting Persons has during the last five years been convicted in a criminal proceeding (excluding traffic violations and similar misdemeanors).
- **(e)** None of the Reporting Persons has during the last five years been a party to a civil proceeding of a judicial or administrative body of competent jurisdiction and as a result of such proceeding was or is subject to a judgment, decree or final order enjoining future violations of, or prohibiting or mandating activities subject to, federal or state securities laws or finding any violation with respect to such laws.
 - **(f)** Mr. Trott is a United States citizen.

Item 3. Source and Amount of Funds or Other Consideration.

The total amount of funds required by the Investor to acquire the shares of Common Stock described in Item 5 is \$680,000,000. All funds in respect of such transaction were drawn from capital contributions of the members of the Investor.

Item 4. Purpose of Transaction

Upon the closing (the "<u>Closing</u>") of the transactions contemplated by the Securities Purchase Agreement, dated as of September 12, 2011, by and among the Investor, the Company, and the other parties thereto (the "<u>Purchase Agreement</u>), the Investor will purchase and acquire (i) 14,756,945 shares (the "<u>Purchased Common Shares</u>") of Common Stock, and (ii) 13,877,552 shares (the "<u>Purchased Preferred Shares</u>") of the Company's Series A Perpetual Convertible Preferred Stock, par value \$0.001 per share ("<u>Preferred Stock</u>"), which will be initially convertible into 12,173,291 shares of Common Stock (the "<u>Underlying Shares</u>").

All of the conditions precedent to the Closing set forth in the Purchase Agreement, other than those the satisfaction of which may be waived in the Investor's sole discretion, have been satisfied as of January 13, 2012, the date on which the High Court of Jersey approved the "Scheme of Arrangement" relating to the Company's previously announced acquisition of Charter International plc ("Charter"). As a result of such satisfaction of all conditions precedent to the Closing, other than those the satisfaction of which may be waived in the Investor's sole discretion, the Investor has the right, subject to no material contingencies, to acquire the Purchased Common Shares, the Purchased Preferred Shares, and the Underlying Shares within sixty (60) days of January 13, 2012. It is anticipated that, after giving effect to other issuances of Common Stock in connection with the Company's acquisition of Charter scheduled to occur on the date of the Closing, there will be 84,693,794 shares of Common Stock issued and outstanding (including the Purchased Common Shares but excluding the Underlying Shares) as of the date of the Closing. Accordingly, as a result of the Investor's beneficial ownership of the Purchased Common Shares and the Underlying Shares, which together comprise 26,930,236 shares of Common Stock, the Investor beneficially owns approximately 27.8% of the aggregate shares of Common Stock.

The Amended and Restated Certificate of Incorporation, which will be adopted by the Company in connection with the Closing, includes a provision that entitles the Investor to nominate up to two (2) individuals for election to the Company's board of directors depending upon the Investor's beneficial ownership of the Company's Common Stock from time to time.

The Reporting Persons intend to hold the Purchased Common Shares and Purchased Preferred Shares, when issued at the Closing, and the Underlying Shares, if and when issued upon conversion of the Purchased Preferred Shares, for investment purposes. However, the Reporting Persons intend to closely monitor the Company's performance and, subject to the covenants and obligations contained in the Purchase Agreement and other agreements pertaining to the Reporting Persons' beneficial ownership of the Common Stock more fully described in Item 6 below, may modify their plans in the future, and may elect to purchase additional shares of Common Stock and/or dispose of shares currently held or hereafter acquired, depending on the Reporting Persons' evaluation of various factors.

Pursuant to the board nomination rights under the Company's Amended and Restated Certificate of Incorporation to be adopted as of the date of the Closing, the Reporting Persons intend to nominate two individuals (who may be principals or officers of the Reporting Persons or their affiliates) to serve as members of the Company's board of directors in connection with the election of directors occurring at the Company's next annual meeting following the date of the Closing. The Reporting Persons also intend to participate in the management of the Company through the exercise contractual consent rights with respect to certain extraordinary transactions involving the Company, including the incurrence of certain indebtedness, as set forth in the Amended and Restated Certificate of Incorporation.

The disclosures set forth in Item 6 below and the Exhibits to this Schedule 13D set forth in Item 7 below are incorporated by reference in this Item 4.

Item 5. Interest in Securities of the Issuer.

- (a)-(b)(i) the Investor is the direct beneficial owner of 26,930,236 shares of Common Stock. This amount is based upon the Purchased Common Shares and Purchased Preferred Shares to be issued to the Investor on the date of the Closing under the Purchase Agreement referred to in Item 4 and the number of shares of Common Stock into which the Purchased Preferred Shares may be converted as of the date of the Closing.
- (ii) Based upon the 84,693,795 shares of Common Stock to be outstanding as of the date of the Closing (including the Purchased Common Shares but excluding the Underlying Shares), the number of shares of Common Stock directly beneficially owned by the Investor represents approximately 27.8% of the aggregate

shares of Common Stock as of the date of the Closing. The foregoing amounts are based upon and assume the completion of the transactions described in the foregoing Item 4 and are based upon information set forth in the Company's definitive proxy statement on Form 14A filed with the Securities and Exchange Commission on December 19, 2011.

- (iii) the Investor, acting through its manager, BDTCP GP I, LLC, will have the sole power to vote or to direct the vote, and to dispose or to direct the disposition of, the shares of Common Stock beneficially owned by it.
- (iv) By virtue of the relationships described under Item 2 of this Schedule 13D, each of the other Reporting Persons may be deemed to share beneficial ownership of the shares of Common Stock directly beneficially owned by the Investor.
- (v) Each of BDT CP, BDTCP GP I and the Investor affirms membership in a group with each other but disclaims membership in a group with BDTP or Mr. Trott. Each of BDTP and Mr. Trott disclaims membership in a group with any person.

The filing of this Schedule 13D shall not be construed as an admission that any Reporting Person is the beneficial owner of any of the shares of Common Stock that such Reporting Person may be deemed to beneficially own. Without limiting the foregoing sentence, each of BDTP and Mr. Trott disclaims beneficial ownership of all shares of Common Stock reported in this Schedule 13D.

- **(c)** The Purchased Common Shares and the Purchased Preferred Shares are being purchased from the Company pursuant to the transactions contemplated by the Purchase Agreement described in Item 4 above. The aggregate purchase price paid for the Purchased Common Shares and Purchased Preferred Shares will be \$680,000,000.
- **(d)** Except as stated elsewhere in this Item 5 or in Item 6, no other person has the right to receive or the power to direct the receipt of dividends from, or the proceeds from the sale of, the shares of Common Stock owned by the Investor.
 - (e) Not applicable.

Item 6. Contracts, Arrangements, Understandings or Relationships With Respect to Securities of the Issuer.

The Investor has the right to acquire the Purchased Common Shares and the Purchased Preferred Shares pursuant to the Purchase Agreement. Among other covenants and agreements, the Purchase Agreement contains:

- (1) A "standstill agreement" restricting the Investor from taking certain actions with respect to mergers, acquisitions and other extraordinary transactions involving the Company, soliciting proxies of the Company's shareholders, becoming a "participant" in any "election contest" (in each case as defined in Rule 14a-11 promulgated under the Securities Exchange Act of 1934), and other similar actions involving the Company;
 - (2) Provisions restricting the ability of the Investor to transfer the Purchased Common Shares, Purchased Preferred Shares and Underlying Shares; and
- (3) "Tag-Along Rights" pursuant to which the Investor has the right to participate in certain sale transactions involving shares of Common Stock initiated by Stephen M. Rales and/or Mitchell P. Rales.

This summary of the Purchase Agreement is qualified in its entirety by reference to the express provisions of such agreement incorporated in this Schedule 13D as detailed in Item 7 of this Schedule 13D.

In connection with the Closing, the Investor and the Company will enter into a Registration Rights Agreement, to be dated as of the date of the Closing (the "Registration Rights Agreement"). The Registration Rights Agreement provides for certain rights relating to registration under the Securities Act of 1933 of the Purchased Common Shares and Underlying Shares, including "shelf", "demand" and "piggy-back" registration rights. This summary of the Registration Rights Agreement is qualified in its entirety by reference to the express provisions of such agreement incorporated in this Schedule 13D as detailed in Item 7 of this Schedule 13D.

In connection with the transactions described above, the members of the Investor and BDTCP GP I, in its capacity as manager of the Investor, have adopted an amended and restated operating agreement of the Investor (the "<u>Investor Operating Agreement</u>"). The Investor Operating Agreement provides that the Investor will distribute to its members *pro rata* (i) 50% of the Purchased Common Shares and/or any cash received upon a disposition thereof by no later than nine (9) months after the date of the Closing, (ii) 100% of the Purchased Common Shares and/or any cash received upon a disposition thereof by no later than fifteen (15) months following the date of the Closing; (iii) 100% of the Purchased Preferred Shares and/or any cash received upon a disposition thereof by no later than five (5) years after the date of the Closing, and (iv) dividends received on the Purchased Common Shares and Purchased Preferred Shares upon receipt thereof. This summary of the Investor Operating Agreement is qualified in its entirety by reference to the express provisions of such agreement incorporated in this Schedule 13D as detailed in Item 7 of this Schedule 13D.

Item 7. Material to Be Filed as Exhibits.

Exhibit No.	Description of Exhibit
1.	Securities Purchase Agreement, dated as of September 12, 2011.*
2.	Form of Registration Rights Agreement.*
3.	Form of Certificate of Designations Series A Perpetual Convertible Preferred Stock.
4.	Form of Amended and Restated Certificate of Incorporation.*
5.	Amended and Restated Limited Liability Company Agreement of BDT CF Acquisition Vehicle, LLC.
6.	Joint Filing Agreement, dated as of January 23, 2012.

^{*} Incorporated by reference to Exhibit 99.2 to the Company's Current Report on Form 8-K filed on September 15, 2011.

SIGNATURES

After reasonable inquiry and to the best of my knowledge and belief, I certify that the information set forth in this statement is true, complete and correct.

Dated: January 23, 2012

BDT CAPITAL PARTNERS, LLC

By: /s/ Byron D. Trott

Name: Byron D. Trott

Its: President and Chief Investment Officer

BDTCP GP I, LLC

By: /s/ Byron D. Trott

Name: Byron D. Trott

President and Chief Investment Officer

BDT CF ACQUISITION VEHICLE, LLC

By: BDTCP GP I, LLC

Its: Manager

By: /s/ Byron D. Trott

Name: Byron D. Trott

President and Chief Investment Officer

BYRON D. TROTT

/s/ Byron D. Trott

BDTP GP, LLC

By: /s/ Byron D. Trott

Name: Byron D. Trott

Its: President and Chief Investment Officer

CERTIFICATE OF DESIGNATIONS OF SERIES A PERPETUAL CONVERTIBLE PREFERRED STOCK

(PAR VALUE \$0.001)

OF

COLFAX CORPORATION

Pursuant to Section 151 of the

General Corporation Law of the State of Delaware

Colfax Corporation, a corporation organized and existing under the General Corporation Law of the State of Delaware (the "Corporation"), in accordance with the provisions of Section 151 thereof, DOES HEREBY CERTIFY:

That pursuant to the authority conferred upon the Board of Directors of the Corporation (the "Board") in accordance with the Second Amended and Restated Certificate of Incorporation of the Corporation (the "Certificate of Incorporation") and the Amended and Restated Bylaws of the Corporation (the "Bylaws"), the Board on September 10, 2011 adopted the following resolution, creating a series of 13,877,552 shares of Preferred Stock, par value \$0.001 per share, of the Corporation designated as Series A Perpetual Convertible Preferred Stock, which is subject to shareholder approval of certain amendments to the Certificate of Incorporation including amendments that will increase the authorized number of shares of preferred stock:

RESOLVED, that pursuant to the authority vested in the Board by Article IV of the Certificate of Incorporation and out of the Preferred Stock, par value \$0.001 per share, authorized therein, the Board hereby authorizes, designates and creates a series of Preferred Stock, and states that the designation and amount thereof and the voting powers, preferences and relative, participating, optional and other special rights of the shares of such series, and the qualifications, limitations or restrictions thereof be, and hereby are, as follows:

Section 1. Number of Shares and Designation.

The designation of the series of Preferred Stock created by this resolution shall be "Series A Perpetual Convertible Preferred Stock" (the "Series"), and the number of shares constituting the Series shall be Thirteen Million Eight Hundred Seventy-Seven Thousand Five Hundred Fifty-Two (13,877,552) (the "Series A Preferred Stock"). Each share of Series A Preferred Stock shall have a liquidation preference of \$24.50 (the "Liquidation Preference").

Section 2. Dividends.

- (1) Holders of Series A Preferred Stock shall be entitled to receive, on each share of Series A Preferred Stock, out of funds legally available for the payment of dividends under Delaware law, cumulative cash dividends with respect to each Dividend Period (as defined below) at a per annum rate of 6% (as such may be adjusted pursuant to this Section 2(1), the "Dividend Rate") on (i) the Liquidation Preference per share and (ii) to the extent unpaid on the Dividend Payment Date (as defined below), the amount of any accrued and unpaid dividends, if any, on such share of Series A Preferred Stock; provided that if, on any Dividend Payment Date, the Corporation shall not have paid in cash the full amount of any dividend required to be paid on such share (such amount being "Unpaid Dividends") on such Dividend Payment Date pursuant to this Section 2(1), then from such Dividend Payment Date, the Dividend Rate shall automatically be at a per annum rate of 8% for such share until the date on which all Unpaid Dividends have been declared and paid in full in cash. Dividends shall begin to accrue and be cumulative from the Issue Date (whether or not declared), shall compound on each Dividend Payment Date, and shall be payable in arrears (as provided below in this Section 2(1)), but only when, as and if declared by the Board (or a duly authorized committee of the Board) on each March 1, June 1, September 1 and December 1, and each Mandatory Conversion Date, Redemption Date and Liquidation Date (each, a "Dividend Payment Date"); provided that if any such Dividend Payment Date would otherwise occur on a day that is not a Business Day, such Dividend Payment Date shall instead be (and any dividend payable on Series A Preferred Stock on such Dividend Payment Date shall instead be payable on) the immediately succeeding Business Day with no additional dividends payable as a result of such payment being made on such succeeding Business Day. Dividends that are payable on Series A Preferred Stock on any Dividend Payment Date will be payable to holders of record of Series A Preferred Stock as they appear on the stock register of the Corporation on the applicable record date, which shall be the fifteenth (15th) calendar day before such Dividend Payment Date (as originally scheduled) or such other record date fixed by the Board (or a duly authorized committee of the Board) that is not more than 60 nor less than 10 days prior to such Dividend Payment Date (each, a "Dividend Record Date"). Any such day that is a Dividend Record Date shall be a Dividend Record Date whether or not such day is a Business Day. Each dividend period (a "Dividend Period") shall commence on and include the calendar day immediately following a Dividend Payment Date (other than the initial Dividend Period, which shall commence on and include the Issue Date) and shall end on and include the next Dividend Payment Date. Dividends payable in respect of a Dividend Period shall be payable in arrears on the first Dividend Payment Date after such Dividend Period.
- (2) The Corporation (including its subsidiaries) shall not declare, pay or set apart funds for any dividends or other distributions with respect to any Junior Stock of the Corporation or repurchase, redeem or otherwise acquire, or set apart funds for repurchase, redemption or other acquisition of, any Junior Stock, or make any guarantee payment with respect thereto, unless all accrued but unpaid dividends on the Series A Preferred Stock for all Dividend Periods through and including the date of such declaration, payment, repurchase, redemption or acquisition (including, if applicable as provided in Section 2(1) above, dividends on such amount) have been declared and paid in full in cash (or declared and a sum sufficient for the payment thereof set apart for such payment). Without limitation of the foregoing, any such dividend or other distributions on the Junior Stock shall be subject to Section 2(3).
- (3) In the event that any dividend is declared and paid on, or any distribution is made with respect to, any Junior Stock (including, without limitation, in connection with a recapitalization of the Corporation), the Series A Preferred Stock shall share proportionately with such Junior Stock in any such dividend or distribution, (a) if such Junior Stock is Common Stock or is convertible into Common Stock, in accordance with the number of shares of Common Stock issuable upon conversion of the Series A Preferred Stock calculated as of the record date for such dividend or

distribution, or (b) if such Junior Stock is not Common Stock or convertible into Common Stock, in such manner and at such time as the Board may determine in good faith to be equitable in the circumstances.

(4) Any reference to "dividends" or "distributions" in this Section 2 shall not be deemed to include any distribution made in connection with any voluntary of involuntary dissolution, liquidation or winding up of the Corporation.

Section 3. Mandatory Conversion; Optional Conversion; Redemption.

- (1) Mandatory Conversion at the Option of the Corporation.
- (a) <u>Mandatory Conversion</u>. On or after the third anniversary of the Issue Date, the Corporation shall have the right, at its option, at any time or from time to time, to cause some or all of the outstanding shares of Series A Preferred Stock to be mandatorily converted (a "<u>Mandatory Conversion</u>") into fully paid and nonassessable shares of Common Stock at the Conversion Rate in effect as of the applicable Mandatory Conversion Date (as defined below) in accordance with the provisions of this Section 3(1), subject to satisfaction of the following conditions:
 - (i) the closing price of the Common Stock on the New York Stock Exchange (or, if the shares of Common Stock are not listed or admitted for trading on the New York Stock Exchange, as reported on the principal consolidated transaction reporting system on the principal national securities exchange on which the shares of Common Stock are listed or admitted for trading) shall exceed one-hundred and thirty-three percent (133%) of the Conversion Price for a period of thirty (30) consecutive trading days ending on the trading day immediately preceding the date that the Corporation delivers a Mandatory Conversion Notice (as defined below); and
 - (ii) the Corporation shall have declared and paid in full in cash, or shall have declared and set apart for payment in cash, all accrued but unpaid dividends (including, if applicable as provided in Section 2(1) above, dividends on such amount) on the Series A Preferred Stock for all Dividend Periods through and including the effective date of such Mandatory Conversion (the "Mandatory Conversion Date").
- (b) Mechanics of Mandatory Conversion. In order to effect a Mandatory Conversion, the Corporation shall provide written notice (a "Mandatory Conversion Notice") to each holder of outstanding shares of Series A Preferred Stock by first class mail, postage prepaid, to such holder at such holder's address as it shall appear in the records of the Corporation or such other address as such holder shall specify to the Corporation in writing from time to time. The Mandatory Conversion Notice shall specify (i) the applicable Mandatory Conversion Date, and (ii) the number of shares of Common Stock that each holder shall be entitled to receive in connection with such Mandatory Conversion. Upon receipt of the Mandatory Conversion Notice, each holder of shares of Series A Preferred Stock shall surrender his, her, or its certificate of certificates for all such shares (or, if applicable, a pro rata portion thereof determined in accordance with the penultimate sentence of this Section

3(1)(b)) (or, if such certificate or certificates have been lost, stolen, or destroyed, a lost certificate affidavit and indemnity in form and substance reasonably acceptable to the Corporation) to the Corporation at its principal office or at the office of the agency which may be maintained for the purpose of administering conversions and redemptions of the shares of Common Stock (the "Conversion Agent"), in each case, as specified in the applicable Mandatory Conversion Notice. All rights with respect to the shares of Series A Preferred Stock that are converted pursuant to a Mandatory Conversion shall terminate on the Mandatory Conversion Date, subject to the rights of the holders thereof to receive the items provided for in the following sentence. As soon as practicable after the Mandatory Conversion Date, but in no event later than ten (10) days after the Mandatory Conversion Date, the Corporation shall issue and deliver to each holder that has surrendered shares of Series A Preferred Stock in connection therewith, (i) a certificate or certificates (or if the holder shall so elect, and if permitted by applicable law, including without limitation the Securities Act of 1933, uncertificated book-entry shares) representing the number of fully paid and nonassessable shares of Common Stock issuable upon such conversion, with no personal liability attaching to the ownership thereof, free of all taxes with respect to the issuance thereof, liens, charges and security interests and not subject to any preemptive rights, into which such shares of Series A Preferred Stock have been converted in connection with such Mandatory Conversion, and (ii) any cash payable in respect of fractional shares as provided in Section 3(4). The converted shares of Series A Preferred Stock shall be retired and cancelled and may not be reissued. In the case of Mandatory Conversion of fewer than all of the shares of Series A Preferred Stock at the time outstanding, the shares of Series A Preferred Stock to be redeemed shall be selected, and allocated among the holders of outstanding shares of Series A Preferred Stock on a pro rata basis in accordance with the number of shares of Series A Preferred Stock then held by each such holder. In the event that fewer than all of the shares of Series A Preferred Stock represented by a certificate are converted in connection with a Mandatory Conversion, then a new certificate representing the unconverted shares of Series A Preferred Stock shall be issued to the holder of such certificate.

(2) Redemption at the Option of the Corporation.

- (a) <u>Redemption</u>. On or after the fifth anniversary of the Issue Date, the Corporation shall have the right, at its option, at any time to redeem (a "<u>Redemption</u>") all (but not less than all) of the outstanding shares of Series A Preferred Stock in return for a payment in cash (the "<u>Call Payment</u>") equal, on a per share basis, to the greater of (i) the Conversion Price, and (ii) the Liquidation Preference, in each case, as of the applicable Redemption Date (as defined below) in accordance with the provisions of this Section 3(2), subject to satisfaction of the following conditions:
 - (i) on the trading date preceding the date of the Redemption Notice (as defined below) the closing price of the Common Stock on the New York Stock Exchange (or, if the shares of Common Stock are not listed or admitted for trading on the New York Stock Exchange, as reported on the principal consolidated transaction reporting system on the principal national securities exchange on which the shares of Common Stock are listed or admitted for trading) shall be less than the Conversion Price; and
 - (ii) the Corporation shall have declared and paid in full in cash, or shall have declared and set apart for payment in cash, all accrued but unpaid dividends (including, if applicable as provided in Section 2(1) above, dividends on such amount) on the Series A Preferred Stock for all Dividend Periods through and including the Redemption Date (as defined below).

(b) Mechanics of Redemption. In order to effect a Redemption, the Corporation shall provide written notice (a "Redemption Notice") of such Redemption to each holder of outstanding shares of Series A Preferred Stock by first class mail, postage prepaid, to such holder at such holder's address as it shall appear in the records of the Corporation or such other address as such holder shall specify to the Corporation in writing from time to time. Subject to the satisfaction of the conditions set forth in Section 3(2)(a)(i) and Section 3(2)(a)(ii), the Redemption shall become effective on the ninetieth (90th) day after delivery of the Redemption Notice, or if such date is not a Business Day, then on the next Business Day (such date, the "Redemption Date"). The Redemption Notice shall specify (i) the applicable Redemption Date (determined in accordance with the preceding sentence), and (ii) the Call Payment to which each holder of outstanding shares of Series A Preferred Stock shall be entitled in connection with such Redemption. On or before the applicable Redemption Date, each holder of outstanding shares of Series A Preferred Stock to be redeemed shall surrender the certificate or certificates representing such shares (or, if such certificate or certificates have been lost, stolen, or destroyed, a lost certificate affidavit and indemnity in form and substance reasonably acceptable to the Corporation) to the Corporation at its principal office or to the Conversion Agent, in each case, as may be specified in the Redemption Notice, and upon receipt thereof by the Corporation or the Conversion Agent, as the case may be, the Call Payment for such redeemed shares shall be immediately due and payable in cash to the order of the record holder of the shares of Series A Preferred Stock being redeemed. From and after the Redemption Date, all dividends on shares of Series A Preferred Stock that are redeemed on such Redemption Date shall cease to accumulate and all rights of the holders thereof as holders of Series A Preferred

(3) Right of Holders to Optionally Convert.

- (a) Each share of Series A Preferred Stock shall be convertible, in whole or in part, at the option of the holder thereof (an "Optional Conversion"), at any time after the Issue Date, and from time to time, and without payment of any additional consideration by the holder of such share, into fully paid and nonassessable shares of Common Stock at the Conversion Rate in effect as of the applicable Optional Conversion Date (as defined below) in accordance with the procedures set forth in this Section 3(3).
- (b) In order to effect an Optional Conversion, the holder of shares of Series A Preferred Stock to be converted shall (i) deliver a properly completed and duly executed written notice of election to convert (an "Optional Conversion Notice") to the Corporation at its principal office or to the Conversion Agent, and (ii) surrender the certificates for the shares of Series A Preferred Stock that are to be converted, accompanied, if so required by the Corporation or the Conversion Agent, by a written instrument or instruments of transfer in form reasonably satisfactory to the Corporation or the Conversion Agent, duly executed by the holder or its attorney duly authorized in writing. The Optional Conversion Notice shall specify: (i) the number (in whole shares) of shares of Series A Preferred Stock to be converted, and (ii) the name or names in which the converting holder wishes the certificates for shares of Common Stock in connection with such conversion to be issued.

- (c) The Optional Conversion shall become effective at the close of business on the date (such date, the "Optional Conversion Date") of receipt by the Corporation or the Conversion Agent of the Optional Conversion Notice and the other items referred to in Section 3(3)(b). Promptly following the Optional Conversion Date (and in no event more than ten (10) days after the Optional Conversion Date), the Corporation shall deliver or cause to be delivered at the office or agency of the Conversion Agent, to or upon the written order of the holders of the surrendered shares of Series A Preferred Stock, (i) a certificate or certificates (or if the converting holder shall so elect, and if permitted by applicable law, including without limitation the Securities Act of 1933, uncertificated book-entry shares) representing the number of fully paid and nonassessable shares of Common Stock issuable upon such conversion, with no personal liability attaching to the ownership thereof, free of all taxes with respect to the issuance thereof, liens, charges and security interests and not subject to any preemptive rights, into which such shares of Series A Preferred Stock have been converted in connection with such Optional Conversion, (ii) a cash payment equal to the amount of all accrued but unpaid dividends (including, if applicable as provided in Section 2(1) above, dividends on such amount) for all Dividend Periods through and including the most recent Dividend Payment Date (the "Cash Payment"), provided, that, if on such Optional Conversion Date, the Corporation has not declared all or any portion of the accrued but unpaid dividends (including, if applicable as provided in Section 2(1) above, dividends on such amount) for all Dividend Periods through and including the most recent Dividend Payment Date, the converting holder shall receive (instead of and in full satisfaction of the Corporation's obligation to make the Cash Payment) such number of shares of Common Stock equal to the amount of accrued but unpaid dividends (including, if applicable as provided in Section 2(1) above, dividends on such amount), divided by the "current market price" per share of Common Stock, and (iii) any cash payable in respect of fractional shares as provided in Section 3(4). Except as described above, upon any Optional Conversion, the Corporation shall make no payment or allowance for unpaid dividends on the shares of Series A Preferred Stock that are converted in connection with such Optional Conversion. In the case of an Optional Conversion Date that occurs after a Dividend Record Date and before a Dividend Payment Date, the Cash Payment shall be reduced in an amount equal to any dividends declared with respect to such Dividend Record Date on any shares of Series A Preferred Stock that are converted on such Optional Conversion Date, provided that such dividends are actually received by the record holder of such shares on such Dividend Payment Date; or, to the extent that such reduction in the Cash Payment is less than the amount of such dividends to be received with respect to such Dividend Record Date, the excess amount thereof will be repaid to the Corporation.
- (d) Upon the surrender of a certificate representing shares of Series A Preferred Stock that is converted in part, the Corporation shall issue or cause to be issued to the surrendering holder a new certificate representing shares of Series A Preferred Stock equal in number to the unconverted portion of the shares of Series A Preferred Stock represented by the certificate so surrendered.
- (e) On the Optional Conversion Date, upon the delivery to the converting holder of the shares Common Stock issuable in connection with such conversion and the payments referred to in Section 3(3)(c)(ii) and (iii), the rights of the holders of the shares of converted Series A Preferred Stock and the person entitled to receive the shares of Common Stock upon the conversion of such shares of Series A Preferred Stock shall be treated for all purposes as having become the Beneficial Owner of such shares of Common Stock.

(4) Fractional Shares.

- (a) No fractional shares or scrip representing fractional shares of Common Stock shall be issued upon the conversion of any shares of Series A Preferred Stock in connection with a Mandatory Conversion or Optional Conversion. Instead of any fractional interest in a share of Common Stock which would otherwise be deliverable upon the conversion of a share of Series A Preferred Stock in connection with a Mandatory Conversion or Optional Conversion, the Corporation shall pay to the holder of such share of Series A Preferred Stock an amount in cash (computed to the nearest cent) equal to the product of (A) such fraction and (B) the current market price (as defined below) of a share of Common Stock on the Business Day next preceding the day of conversion. If more than one share of Series A Preferred Stock shall be surrendered for conversion at one time by the same holder, the number of full shares of Common Stock issuable upon conversion thereof shall be computed on the basis of the aggregate Liquidation Preference of the shares of Series A Preferred Stock so surrendered.
- (b) For the purposes of this Section 3, the "current market price" per share of Common Stock at any date shall be deemed to be the average of the daily closing prices on the New York Stock Exchange (or if the shares of Common Stock are not listed or admitted for trading on the New York Stock Exchange as reported on the principal consolidated transaction reporting system on the principal national securities exchange on which the shares of Common Stock are listed or admitted for trading) for the ten consecutive trading days immediately prior to the date in question or in the event that no trading price is available for the shares of Common Stock, the fair market value thereof, as determined in good faith by the Board.
- (5) <u>Reservation of Shares</u>. The Corporation shall at all times when the Series A Preferred Stock shall be outstanding, reserve and keep available out of its authorized but unissued Common Stock, for the purpose of effecting the conversion of the Series A Preferred Stock pursuant to Section 3(1) and Section 3(3), such number of its duly authorized shares of Common Stock as shall from time to time be sufficient to effect the conversion of all outstanding Series A Preferred Stock.
- (6) <u>Certain Events</u>. If the Corporation shall be a party to any transaction, including without limitation a merger, consolidation, sale of all or substantially all of the Corporation's assets, reorganization, liquidation or recapitalization of the Common Stock (each of the foregoing being referred to as a "<u>Transaction</u>"), in each case as a result of which shares of Common Stock shall be converted into the right to receive stock, securities or other property (including cash or any combination thereof), each share of Series A Preferred Stock shall, upon consummation of the Transaction, be convertible into the kind and amount of shares of stock and other securities and property receivable (including cash) upon the consummation of such Transaction by a holder of that number of shares of Common Stock into which one share of Series A Preferred Stock was convertible immediately prior to such Transaction.

Section 4. Voting Rights.

(1) Each share of Series A Preferred Stock shall entitle the holder thereof to vote on all matters voted on by holders of the Common Stock, voting together as a single class with the shares of Common Stock entitled to vote, at all meetings of the stockholders of the Corporation and in connection with any actions of the stockholders of the Corporation taken by written consent. With

respect to any such vote, the holder of each share of Series A Preferred Stock shall be entitled to cast the number of votes equal to the number of votes which could be cast in such vote by a holder of the shares of Common Stock of the Corporation into which such holder's shares of Series A Preferred Stock are convertible as of the record date for such vote.

- (2) The affirmative vote or consent of more than fifty percent (50%) of the shares of Series A Preferred Stock, voting separately as a class, shall be necessary for authorizing, approving, effecting or validating:
 - (a) the amendment, alteration or repeal of any of the provisions of the Certificate of Incorporation or Bylaws or any document amendatory or supplemental thereto (including this Certificate of Designations), whether by merger, consolidation or otherwise, that would adversely affect or cause to be terminated the powers, designations, preferences and other rights of the Series; or
 - (b) any other action for which a vote of the Series A Preferred Stock, voting separately as a class, is required by law.

Other than as specifically set forth in this Section 4 or to the extent of applicable law, the Series A Preferred Stock shall not be entitled to a separate vote on any matter.

Section 5. Liquidation Rights.

- (1) Upon any voluntary or involuntary liquidation, dissolution or winding-up of the Corporation, each holder of shares of the Preferred Stock shall be entitled to payment out of the assets of the Corporation legally available for distribution of an amount per share of Series A Preferred Stock (the "Liquidation Amount") held by such holder equal to the greater of (i) the Liquidation Preference per share of Series A Preferred Stock held by such holder, plus any accrued but unpaid dividends (including, if applicable as provided in Section 2(1) above, dividends on such amount) in respect of such shares, whether or not declared paid in cash, calculated to the date fixed for liquidation, dissolution or winding-up (the "Liquidation Date"), and (ii) the amount that the holders of Series A Preferred Stock would have received in connection with such liquidation, dissolution or winding up had each share of Series A Preferred Stock been converted into shares of Common Stock pursuant to Section 3(3) immediately prior to such liquidation, dissolution or winding up of the Corporation, before any distribution is made on any Junior Stock of the Corporation. After payment in full of the Liquidation Amount to which holders of Series A Preferred Stock are entitled, such holders shall not be entitled to any further participation in any distribution of assets of the Corporation in respect of their shares of Series A Preferred Stock. If, upon any voluntary or involuntary liquidation, dissolution or winding-up of the Corporation, the amounts payable with respect to the Series A Preferred Stock are not paid in full, the holders of the Series A Preferred Stock will share in any distribution of assets of the Corporation on a *pro rata* basis in proportion to the number of shares of Series A Preferred Stock held by each such holder.
- (2) Neither the voluntary sale, conveyance, exchange or transfer (for cash, shares of stock, securities or other consideration) of all or substantially all of the property or assets of the Corporation nor the consolidation or merger of the Corporation with or into one or more Persons will be deemed to be a voluntary or involuntary liquidation, dissolution or winding-up of the Corporation for purposes of this Section 5.

Section 6. Pre-emptive Rights

- (1) Except for (i) (A) the issuance of any shares of Common Stock or options or rights to purchase those shares pursuant to any present or future employee, director or consultant benefit plan, program or practice of or assumed by the Corporation or any of its subsidiaries or (B) the issuance of any shares of Common Stock pursuant to any option, warrant, right or exercisable, exchangeable or convertible security outstanding as of the Issue Date, (ii) a subdivision (including by way of a stock dividend) of the outstanding shares of Common Stock into a larger number of shares of Common Stock, and (iii) the issuance of capital stock as full or partial consideration for a merger, acquisition, joint venture, strategic alliance, or other similar non-financing transaction, if, from the Issue Date until the date that is twenty-four (24) months after the Issue Date, the Corporation wishes to issue any shares of capital stock or any other securities convertible into or exchangeable for capital stock of the Corporation (collectively, "New Securities") at a price per share less than the Liquidation Preference (a "Dilutive Issuance") to any person (the "Proposed Purchaser"), then the Corporation shall send written notice (the "New Issuance Notice") to the holders of the Series A Preferred Stock, which New Issuance Notice shall state (x) the number of New Securities proposed to be issued and (y) the proposed purchase price per share of the New Securities (the "Proposed Price").
- (2) For a period of fifteen (15) Business Days after the giving of the New Issuance Notice, each holder of the Series A Preferred Stock shall have the right to purchase (a) if the Dilutive Issuance occurs during the period from the Issue Date until the date that is two hundred seventy (270) days after the Issue Date, up to its Proportionate Share (as defined below) of the New Securities, or (b) if the Dilutive Issuance occurs after such two-hundred-seventy-(270)-day period, up to its Double Proportionate Share (as defined below) of the New Securities, in each case at a purchase price per share equal to the Proposed Price and upon the terms and conditions set forth in the New Issuance Notice. As used herein, the term "Proportionate Share" means, as to each holder of Series A Preferred Stock as of a given date, the fraction, expressed as a percentage, determined by dividing (i) a number of shares of Common Stock Beneficially Owned by such holder and (without duplication) its Permitted Transferees (other than any other holder of shares of Series A Preferred Stock) plus the number of shares of Common Stock into which the shares of Series A Preferred Stock then Beneficially Owned by such holder are convertible as of such date, by (ii) the number of shares of Common Stock of the Corporation issued and outstanding as of such date, calculated on a fully-diluted basis, assuming conversion or exercise price, but excluding any Unexercised Options and RSUs. As used herein, the term "Double Proportionate Share" means, as to each holder of Series A Preferred Stock as of a given date, such holder's Proportionate Share as of such date multiplied by two (2).
- (3) The right of each holder of Series A Preferred Stock to purchase the New Securities shall be exercisable by delivering written notice of its exercise, prior to the expiration of the fifteen (15) Business Day period referred to in Section 6(2) above, to the Corporation, which notice shall state the amount of New Securities that the holder elects to purchase. The failure of such holder to respond within the fifteen (15) Business Day period or to pay for such New Securities when such payment is due shall be deemed to be a waiver of such holder's rights under this Section 6 only with respect to the Dilutive Issuance described in the applicable New Issuance Notice.
- (4) Except for the foregoing and to the extent arising under applicable law, there are no pre-emptive rights associated with the Series A Preferred Stock.

Section 7. Ranking and Issuance of Parity Stock or Senior Stock Prior to Conversion

The Series A Preferred Stock shall rank senior to any series of Junior Stock with respect to the payment of dividends and distributions and rights upon liquidation, dissolution or winding up of the Corporation. Prior to the conversion or redemption or retirement and cancellation of all shares of the Series A Preferred Stock in accordance with this Certificate of Designations, the Corporation shall not issue an Parity Stock or Senior Stock, or authorize any additional shares of Series A Preferred Stock.

Section 8. Retirement.

If any share of the Series A Preferred Stock is purchased or otherwise acquired by the Corporation in any manner whatsoever, then such share shall be retired and promptly cancelled. Upon the retirement or cancellation of a share of Series A Preferred Stock, such share shall not for any reason be reissued as shares of the Series.

Section 9. Definitions.

Capitalized terms not otherwise defined in this Certificate of Designations shall have the following meanings:

"Beneficial Ownership" and "Beneficially Own" and similar terms shall have the meaning set forth in Rule 13d-3 under the Securities Exchange Act of 1934.

"Board" shall have the meaning set forth in the Preamble.

"Business Day" means a day other than Saturday, Sunday or a day on which banking institutions in any of London, UK, St. Helier, Jersey and New York, New York, USA are authorized or obligated to close.

"Bylaws" shall have the meaning set forth in the Preamble.

"Call Date" shall have the meaning set forth in Section 3(2).

"Call Payment" shall have the meaning set forth in Section 3(2).

"Certificate of Incorporation" shall have the meaning set forth in the Preamble.

"Common Stock" means the Corporation's common stock, par value \$0.001 per share.

"Conversion Agent" shall have the meaning set forth in Section 3(1)(b).

"Conversion Price" means, for each share of Series A Preferred Stock, the U.S. dollar amount equal to the Liquidation Preference divided by the then-applicable Conversion Rate.

"Conversion Rate" means the Liquidation Preference divided by 114% of the Liquidation Preference, subject to the following adjustments:

If the Corporation shall (1) declare or pay a dividend on its outstanding Common Stock in shares of Common Stock or make a distribution to holders of its Common Stock in shares of Common Stock, (2) subdivide its outstanding shares of Common Stock into a greater number of shares of Common Stock, (3) combine its outstanding shares of Common Stock into a smaller number of shares of Common Stock or (4) issue by reclassification of its shares of Common Stock other securities of the Corporation, then the Conversion Rate in effect immediately prior thereto shall be adjusted so that a holder of any shares of Series A Preferred Stock thereafter converted shall be entitled to receive the number and kind of shares of Common Stock or other securities that such holder of Series A Preferred Stock would have owned or been entitled to receive after the happening of any of the events described above had such shares of Series A Preferred Stock been converted immediately prior to the happening of such event or any record date with respect thereto. An adjustment so made shall become effective on the date of the dividend payment, subdivision, combination or issuance retroactive to the record date with respect thereto, if any, for such event. Such adjustment shall be made successively.

- "Corporation" shall have the meaning set forth in the Preamble.
- "Dilutive Issuance" shall have the meaning set forth in Section 6.
- "Dividend Payment Date" shall have the meaning set forth in Section 2(1).
- "Dividend Period" shall have the meaning set forth in Section 2(1).
- "Dividend Rate" shall have the meaning set forth in Section 2(1).
- "Dividend Records Date" shall have the meaning set forth in Section 2(1).
- "Double Proportionate Percentage" shall have the meaning set forth in Section 6.
- "Issue Date" means the date on which any shares of Series A Preferred Stock are first issued by the Corporation.
- "Junior Stock" means the Common Stock and any other class or series of stock of the Corporation, other than Series A Preferred Stock and Senior Stock not expressly ranking senior to the Series A Preferred Stock with respect to the payment of dividends and distributions.

- "Liquidation Date" shall have the meaning set forth in Section 5.
- "Liquidation Amount" shall have the meaning set forth in Section 5.
- "Liquidation Preference" shall have the meaning set forth Section 1.
- "Mandatory Conversion" shall have the meaning set forth in Section 3(1).
- "Mandatory Conversion Date" means the effective date of a mandatory conversion of the shares of Series A Preferred Stock pursuant to Section 3(1).
 - "Mandatory Conversion Notice" shall have the meaning set forth in Section 3(1).
 - "New Issuance Notice" shall have the meaning set forth in Section 6.
 - "New Securities" shall have the meaning set forth in Section 6.
 - "Optional Conversion" shall have the meaning set forth in Section 3(3).
 - "Optional Conversion Date" means the effective date of an optional conversion of the shares of Series A Preferred Stock pursuant to Section 3(3).
 - "Optional Conversion Notice" shall have the meaning set forth in Section 3(3).
- "<u>Parity Stock</u>" means any class or series of stock of the Corporation that would expressly rank equal to the Series A Preferred Stock with respect to the payment of dividends and distributions.
 - "Permitted Transferee" shall have the meaning set forth in the Certificate of Incorporation.
- "<u>Person</u>" means an individual, a partnership, a joint venture, a corporation, a limited liability company, a trust, an unincorporated organization and a government or any department or agency thereof.
 - " $\underline{\text{Preferred Stock}}\text{" means the Corporation's preferred stock, par value $0.001 per share.}$
 - "Proportionate Percentage" shall have the meaning set forth in Section 6.

- "Proposed Purchase" shall have the meaning set forth in Section 6.
- "Redemption" shall have the meaning set forth in Section 3(2).
- "Redemption Date" means the effective date of a redemption of the shares of Series A Preferred Stock pursuant to Section 3(2).
- "Redemption Notice" shall have the meaning set forth in Section 3(2).
- "Senior Stock" means any class or series of stock of the Corporation that would expressly rank senior to the Series A Preferred Stock with respect to the payment of dividends and distributions and/or rights upon liquidation or winding up of the Corporation.
 - "Series" shall have the meaning set forth in Section 1.
 - "Series A Preferred Stock" shall have the meaning set forth in Section 1.
- "<u>Unexercised Options and RSUs</u>" means (i) any convertible securities outstanding as of the Issue Date, or (ii) any options, restricted stock units or similar awards issued to officers, employees, consultants or directors of the Corporation or its subsidiaries, in each case, in the form of Common Stock or entitling the holder thereof to receive shares of Common Stock upon the exercise or conversion thereof.

"Unpaid Dividends" shall have the meaning set forth in Section 2(1).

Section 10. Descriptive Headings and Governing Law.

The descriptive headings of the several Sections and paragraphs of this Certificate of Designations are inserted for convenience only and do not constitute a part of this Certificate of Designations. The General Corporation Law of the State of Delaware shall govern all issues concerning this Certificate of Designations.

IN WITNESS WHEREOF, Colfax Corporation has caused this Certificate to be duly executed in its corporate name this _____ day of ______, 20_____.

COLFAX CORPORATION

By: Name: Title:

AMENDED AND RESTATED LIMITED LIABILITY COMPANY AGREEMENT

OE

BDT CF ACQUISITION VEHICLE, LLC

This **AMENDED AND RESTATED LIMITED LIABILITY COMPANY AGREEMENT** (as amended or amended and restated from time to time, this "<u>Agreement</u>") of BDT CF Acquisition Vehicle, LLC, a limited liability company formed under the laws of the State of Delaware ("<u>CF LLC</u>"), is executed and delivered and made as of January 23, 2012 by and among (i) BDTCP GP I, LLC, a limited liability company formed under the laws of the State of Delaware, as Manager (the "<u>Manager</u>"), and (ii) BDT Capital Partners CF Fund I, L.P., a Delaware limited partnership (the "<u>Co-Investment Fund</u>"), BDT Capital Partners Fund I, L.P., a Delaware limited partnership (the "<u>Parallel Fund</u>"), as Members (collectively, the "<u>Members</u>").

RECITALS

WHEREAS, the parties hereto desire to continue CF LLC as a limited liability company in accordance with the Delaware Limited Liability Company Act, 6 <u>Del. C.</u> § 18-101 <u>et seq.</u> (as in effect on the date of this Agreement and as it may be amended hereafter from time to time, and any successor statute thereto, the "<u>Act</u>") for the purposes specified herein.

NOW, THEREFORE, in consideration of the mutual covenants and promises contained herein and for other good and valuable consideration the receipt and adequacy of which are hereby acknowledged, the parties hereto agree as follows:

ARTICLE 1 DEFINITIONS

As used in this Agreement, the following terms have the meanings set forth below:

"Act" has the meaning set forth in the recitals hereto.

- "Acquisition Agreement" means that certain Securities Purchase Agreement, dated as of September 12, 2011, by and between CF LLC and the Co-Investment Company, as it may be amended from time to time.
- "Adjusted Capital Account" means, with respect to any Member at any time, the balance in such Member's Capital Account, after giving effect to the following adjustments:
 - (a) increase such Capital Account by any amounts which such Member is obligated to restore pursuant to this Agreement or is deemed to be obligated to restore pursuant to Reg. § 1.704-1(b)(2)(ii)(c) or the penultimate sentence of each of Reg. §§ 1.704-2(i)(5) and (g)(1); and

(b) decrease such Capital Account by the items described in Reg. § 1.704-1(b)(2)(ii)(d)(4), (5) and (6).

This definition of Adjusted Capital Account is intended to comply with Reg. § 1.704-1(b)(2)(ii)(d) and shall be interpreted consistently therewith.

"Affiliate" means, with respect to a specified Person, any Person that, directly or indirectly, controls, is controlled by, or is under common control with, the specified Person and shall be deemed to include such Person's Immediate Family and any Family Company or Affiliate thereof. For all purposes of this Agreement, each Principal shall be deemed to be an Affiliate of the Manager for so long as such Principal remains in such capacity. Notwithstanding anything to the contrary herein, for purposes of any restriction on or affecting the activities of Affiliates in this Agreement: (a) none of the BDT Funds nor any portfolio company of any BDT Fund shall be deemed to be an Affiliate of the Manager, any Principal or any of their respective Affiliates; (b) no member of the Fund Advisory Committee or the Senior Advisory Council, as such, shall be deemed to be an Affiliate of the Manager or any Principal; (c) neither BDT & Company, LLC nor any other entity principally engaged in the BDT Advisory Business, shall be deemed to be an Affiliate of the Manager or any Principal; and (d) the Manager shall not be deemed to be in violation or contravention of any such restriction by virtue of any activities attributable to the BDT Advisory Business.

"Assignee" means any Person to which an Interest is Transferred in accordance with Article 9, but who has not been admitted as a Member hereunder.

"Available Assets" means all cash and Securities held by CF LLC (excluding Capital Contributions to be applied to the Co-Investment or CF LLC Expenses), after deducting amounts determined by the Manager to be necessary for the payment of CF LLC Expenses, payments required to be made in connection with liabilities and obligations of CF LLC, capital expenditures and any other amounts set aside for the restoration, increase or creation of reasonable reserves in accordance with the terms of this Agreement.

"Bankruptcy" means, with respect to any Person, the occurrence of any event specified in Section 18-304 of the Act with respect to such Person.

"Business Day" means any weekday, excluding any legal holiday observed pursuant to United States federal or Illinois state, law or regulation.

"Capital Account" means the Capital Account maintained for each Member on CF LLC's books and records in accordance with the following provisions:

(a) To each Member's Capital Account there shall be added (i) such Member's Capital Contributions; (ii) such Member's allocable share of Net Income and any items in the nature of income or gain that are specially allocated to such Member hereunder; and (iii) the amount of any CF LLC liabilities assumed by such Member or which are secured by any property distributed to such Member.

- (b) From each Member's Capital Account there shall be subtracted (i) the amount of (A) cash and (B) the Gross Asset Value of any CF LLC assets (other than cash) distributed to such Member (other than any payment of principal and/or interest to such Member pursuant to the terms of a loan made by such Member to CF LLC); (ii) such Member's allocable share of Net Loss and any other items in the nature of expenses or losses that are specially allocated to such Member hereunder; and (iii) liabilities of such Member assumed by CF LLC or which are secured by any property contributed by such Member to CF LLC.
- (c) In the event any Interest in CF LLC is Transferred in accordance with the terms of this Agreement, the transferee shall succeed to the Capital Account of the transferror to the extent it relates to the Transferred Interest.
- (d) In determining the amount of any liability for purposes of clauses (a) and (b) above, there shall be taken into account Code Section 752(c) and any other applicable provisions of the Code and the regulations thereunder.
- (e) The provisions of this Agreement relating to Capital Account maintenance are intended to comply with Reg. §§ 1.704-1(b) and 1.704-2 and shall be interpreted and applied in a manner consistent therewith. In the event that the Manager shall determine that it is prudent, in order to comply with such regulations, to modify the manner in which the Capital Accounts, or any additions or subtractions thereto, are computed or to maintain equality between the Capital Accounts and the amount of CF LLC capital reflected for book purposes on CF LLC's balance sheet, the Manager may make such modification.

"Capital Contribution" means, with respect to any Member at any time, the aggregate amount of cash contributed to CF LLC by such Member as of such time.

"CF LLC Expenses" means reasonable out-of-pocket costs and expenses of or for (a) maintaining the organizational existence of CF LLC; (b) CF LLC's proportionate share of the fees and expenses of administrators, custodians, outside counsel, consultants, accountants, investment bankers and other similar outside advisors; (c) CF LLC's proportionate share of the costs and expenses incurred in respect of the sourcing, investigating, identifying, analyzing, pursuing, negotiating, consummating, acquiring, financing, holding, monitoring, managing and selling the Co-Investment; (d) CF LLC's proportionate share of the costs and expenses incurred in respect of holding, managing or selling the Co-Investment, including record-keeping expenses; (e) financial, tax and other reporting to the Members and Fund LPs; (f) any taxes, fees or other governmental charges of CF LLC; (g) all other costs, obligations, liabilities and expenses of CF LLC (whether paid directly by CF LLC or by the Manager, subject to reimbursement pursuant to Section 6.6) in connection with this Agreement (such as costs of insurance, costs of litigation, or other matters that are the subject of indemnification or contribution pursuant to Section 6.5, and costs of winding-up and liquidating CF LLC but excluding Manager Expenses); (h) all other incremental costs and expenses of CF LLC's operations (including without limitation, all expenses of CF LLC that correspond with similar expenses incurred by the Main Fund as "Fund Expenses" under the Main Fund Agreement in respect of the Co-Investment), and (i) CF LLC's proportionate share of the costs and expenses incurred in respect of any other costs and expenses jointly incurred by or on behalf of CF LLC, the Co-Investment Fund, the Main Fund, the Parallel Fund and/or any other "Parallel Fund(s)"

(as defined in the Main Fund Agreement) and not otherwise separately allocable among them. To the extent CF LLC Expenses (and comparable expenses in respect of any BDT Funds) are incurred jointly or otherwise in connection with actions intended to benefit CF LLC and any other BDT Fund(s), then the Manager may allocate CF LLC Expenses and such comparable expenses among them to the extent applicable, based on relative investments made and/or benefits derived thereby.

- "Code" means the Internal Revenue Code of 1986, as previously or hereafter amended (or any successor law).
- "Co-Investment" means CF LLC's investment in the Co-Investment Company.
- "Co-Investment Company" means Colfax Corporation and any successor thereto.
- "Co-Investment Fund Agreement" means the Amended and Restated Limited Partnership of the Co-Investment Fund, dated as of January ___, 2012 (as amended and as in effect on the date hereof).
- "Co-Investment Proceeds" means (a) dividends and any similar distributions from the Co-Investment Company, and (b) the net proceeds from any sale, exchange, financing, refinancing, reorganization, liquidation, or other disposition, in whole or in part, of the Common Investment Assets, the Preferred Investment Assets, or any other Securities of the Co-Investment Company.
- "Common Investment Assets" means the Purchased Common Shares (as defined in the Acquisition Agreement) and any other Securities, or other assets, or property received by CF LLC in respect thereof upon any conversion or reclassification thereof, any merger or consolidation of the Co-Investment Company with or into any other Person, any exercise of pre-emptive rights in respect of the Common Investment Assets, or any similar transaction.
- "Depreciation" means an amount equal to the federal income tax depreciation, amortization or other cost recovery deduction allowable with respect to an asset for any Fiscal Year or other period, except that if the Gross Asset Value of an asset differs from its adjusted basis for federal income tax purposes at the beginning of such year or other period, Depreciation shall be an amount that bears the same ratio to such beginning Gross Asset Value as the federal income tax depreciation, amortization or other cost recovery deduction for such year or other period bears to such beginning adjusted tax basis; provided, however, that if the federal income tax depreciation, amortization or other cost recovery deduction for such year or other period is zero, Depreciation shall be determined with reference to such beginning Gross Asset Value using any reasonable method selected by the Manager.
- "Encumbrance" means a charge, pledge, alienation, mortgage, hypothecation, encumbrance or similar collateral assignment or grant of security by any other means, whether for value or no value and whether voluntary or involuntary (including by operation of law or by judgment, levy, attachment, garnishment, winding-up, bankruptcy or other legal or equitable proceedings).

"Engagement Letter" means that certain letter agreement, dated August 17, 2011, by and between the Co-Investment Company and BDT & Company, LLC.

"ERISA" means Title I of the U.S. Employee Retirement Income Security Act of 1974, as previously or hereafter amended.

"Family Company" means any trust (including a grantor trust under Code Sections 671 through 679), partnership, limited liability company, foundation, charitable organization, or other estate-planning vehicle for the benefit of, or the ownership interests of which are owned wholly by, one or more Persons who are Immediate Family members of each other.

"Fund LPs" means, as to each Member, the "Limited Partners" of such Member (as defined in the Co-Investment Fund Agreement, the Main Fund Agreement, or the Parallel Fund Agreement, as applicable).

"Gross Asset Value" means, with respect to any asset, the asset's adjusted basis for federal income tax purposes, except as follows:

- (a) The initial Gross Asset Value of any asset contributed by a Member to CF LLC shall be the gross Fair Value of such asset, as determined under Section 6.4.
- (b) The Gross Asset Values of all CF LLC assets immediately prior to the occurrence of any event described in subparagraphs (i) through (v) below shall be adjusted to equal their respective gross Fair Values, as determined under Section 6.4, upon:
 - (i) the acquisition of an Interest in CF LLC by a new or existing Member in exchange for more than a *de minimis* Capital Contribution, if the Manager reasonably determines that such adjustment is necessary or appropriate to reflect the relative economic interests of the Members;
 - (ii) the liquidation of CF LLC within the meaning of Reg. § 1.704-1(b)(2)(ii)(g), other than a termination of CF LLC under Section 708(b)(1) (B) of the Code;
 - (iii) the distribution to a Member of more than a *de minimis* amount of CF LLC assets as consideration for an Interest, if the Manager reasonably determines that such adjustment is necessary or appropriate to reflect the Members' relative economic interests;
 - (iv) the grant of an Interest (other than a *de minimis* interest) as consideration for the provision of services to or for the benefit of CF LLC by an existing Member acting in a member capacity, or by a new Member acting in a member capacity or in anticipation of becoming a member of CF LLC, if the Manager reasonably determines that such adjustment is necessary or appropriate to reflect the Members' relative economic interests; and

- (v) at such other times as the Manager shall reasonably determine to be necessary or advisable in order to comply with Reg. §§ 1.704-1(b) and 1.704-2.
- (c) The Gross Asset Value of any CF LLC asset distributed to a Member shall be the gross Fair Value of such asset on the date of distribution as determined under Section 6.4.
- (d) The Gross Asset Values of CF LLC assets shall be increased or decreased as necessary to reflect any adjustments to the adjusted basis of such assets pursuant to Code Sections 734(b) or 743(b), but only to the extent that such adjustments are taken into account in determining Capital Account amounts under Reg. § 1.7041(b)(2)(iv)(m); provided, however, that Gross Asset Values shall not be adjusted pursuant to this clause (d) to the extent that an equivalent adjustment is made under clause (b).
- (e) If the Gross Asset Value of a CF LLC asset has been determined or adjusted under clause (a), (b) or (d) above, such Gross Asset Value shall thereafter be adjusted by the Depreciation taken into account with respect to such CF LLC asset for purposes of computing Net Income and Net Loss.
- "Immediate Family" means (a) for any natural person, such person's current spouse, parents, parents-in-law, grandparents, children (including any adopted children), siblings or grandchildren and/or any Family Company, and (b) for any Family Company, the Immediate Family (as defined in clause (a)) of any beneficial owner of such Family Company.
- "Incapacity" means, with respect to any Person, the entry of an order of incompetence or of insanity, or the permanent physical incapacity, death, dissolution, withdrawal, Bankruptcy or termination (other than by merger or consolidation) of such Person.
- "Interest" or "Membership Interest" means the entire limited liability company interest of a Member in CF LLC at any particular time, including such Member's right to share in Net Income, Net Loss, or items thereof, and to receive distributions from, CF LLC, any and all rights to vote, and the rights to any and all benefits to which such Member is entitled hereunder, together with the obligation of such Member to comply with all of the terms and provisions hereof.
 - "Investment Advisers Act" means the United States Investment Advisers Act of 1940, as previously or hereafter amended.
 - "Investment Company Act" means the United States Investment Company Act of 1940, as previously or hereafter amended.
- "Main Fund Agreement" means the Third Amended and Restated Limited Partnership of the Main Fund, dated as of June 29, 2010 (as amended and as in effect on the date hereof).
- "Majority [or other percentage] in Interest" means, with respect to Members, at any time, Members holding more than 50% (or such other applicable percentage) of the aggregate Percentage Interests held by all such Members; provided, however, that, as to any matter

requiring the consent or approval of a Majority (or other applicable percentage) in Interest of the Members hereunder, each Member shall be deemed to have given such consent or such approval only if such Member shall have obtained the consent or approval of the Fund LPs of such Member under the corresponding provisions of the Co-Investment Agreement, the Main Fund Agreement, or the Parallel Fund Agreement, as applicable.

- "Malfeasance" means, with respect to any Person, any act or omission that constitutes: (a) gross negligence in respect of CF LLC or any investment thereof; (b) fraud or willful misconduct; or (c) a willful breach of this Agreement that remains uncured for at least 30 days after the Manager had actual knowledge, or should have had knowledge, thereof.
- "Manager" has the meaning assigned to such term in the Preamble to this Agreement. The Manager shall be the sole "manager" of CF LLC within the meaning of the Act.
- "Manager Expenses" means the general overhead expenses of the Manager, including salaries, bonuses and benefits of employees of the Manager, rent, entertainment, travel, office furniture, fixtures and computer equipment; *provided* that, the foregoing shall not include CF LLC Expenses.
- "Marketable Securities" means Securities that are, at any relevant time, traded on a U.S. national securities exchange, a major non-U.S. exchange or reported through the National Association of Securities Dealers Automated Quotation System or comparable non-U.S. established over-the-counter trading system, or otherwise traded over-the-counter.
- "Member" has the meaning assigned to such term in the Preamble to this Agreement. For purposes of the Act, the Members of CF LLC shall constitute a single class or group of "members".
- "Net Income" or "Net Loss" means an amount equal to CF LLC's taxable income or loss, as the case may be, for a Fiscal Year or other period, determined in accordance with Code Section 703(a) (for this purpose, all items of income, gain, loss or deduction required to be stated separately pursuant to Code Section 703(a)(1) shall be included in taxable income or loss, as applicable), with the following adjustments:
 - (a) Any income of CF LLC that is exempt from federal income tax and not otherwise taken into account in computing Net Income or Net Loss shall increase the amount of such income and/or decrease the amount of such loss;
 - (b) Any expenditures of CF LLC described (or treated under Reg. \S 1.704-1(b)(2)(iv)(i) as described) in Code Section 705(a)(2)(B) and not otherwise taken into account in computing Net Income or Net Loss shall decrease the amount of such income and/or increase the amount of such loss;
 - (c) In the event the Gross Asset Value of any CF LLC asset is adjusted pursuant to clause (b) or clause (c) of the definition of Gross Asset Value, the amount of such adjustment shall be taken into account in the taxable year of such adjustment as gain or loss from the disposition of such asset for purposes of computing Net Income or Net Loss;

- (d) Gain or loss resulting from any disposition of property with respect to which gain or loss is recognized for federal income tax purposes shall be computed by reference to the Gross Asset Value (and not the adjusted tax basis) of the property disposed of;
 - (e) In lieu of federal income tax depreciation, amortization and other cost recovery deductions, there shall be taken into account Depreciation;
- (f) To the extent an adjustment to the adjusted tax basis of any CF LLC asset pursuant to Code Sections 734(b) or 743(b) is required under Reg. § 1.7041(b)(2)(iv)(m)(4) to be taken into account in determining Capital Accounts as a result of a nonliquidating distribution, the amount of such adjustment shall be treated as an item of gain (if the adjustment increases the basis of the asset) or loss (if the adjustment decreases the basis of the asset) from the disposition of the asset and shall be taken into account in computing Net Income or Net Loss; and
- (g) Notwithstanding any other provision of this definition of Net Income and Net Loss, any items specially allocated under Article 5 hereof shall not be taken into account in computing Net Income or Net Loss, but shall be determined by applying rules analogous to those set forth in this definition.
- "Parallel Fund Agreement" means the Second Amended and Restated Limited Partnership Agreement of the Parallel Fund, dated as of June 29, 2010 (as amended and as in effect on the date hereof).
- "Percentage Interest" means, with respect to any Member (as such), or any subset thereof, that percentage which corresponds to the ratio that the aggregate amount of each such Member's Capital Contributions pursuant to Section 3.2.2 bears to the total then-current aggregate amount of such Capital Contributions made by all Member, or subset thereof.
- "Person" means and includes an individual, a partnership, a limited liability company, a joint venture, a corporation, a trust, an unincorporated organization, a government or any department or agency thereof or any entity similar to any of the foregoing.
- "Placement Agent Agreement" means that certain letter agreement, dated as of September 1, 2011, by and between the Co-Investment Company and BDT & Company, LLC.
- "Preferred Investment Assets" means the Purchased Preferred Shares (as defined in the Acquisition Agreement) and any other Securities, or other assets, or property received by CF LLC in respect thereof upon any conversion or reclassification thereof, any merger or consolidation of the Co-Investment Company with or into any other Person, any exercise of pre-emptive rights in respect of the Preferred Investment Assets, or any similar transaction.
- "Reg. §" or "Reg. §" means a temporary or final regulation or regulations promulgated by the U.S. Treasury Department under the Code, as such regulations may be amended from time to time.

"Securities" means any of one or more of the following: (a) capital stock (both common and preferred); partnership interests (both limited and general); limited liability company interests; interests in any acquisition, venture capital or other investment funds; notes; bonds; debentures; other obligations, instruments or evidences of indebtedness (whether convertible or otherwise); and other securities and equity interests of whatever kind of any Person, whether readily marketable or not; (b) any rights to acquire and/or sell any of the Securities described in clause (a) above (including calls, puts, options, warrants, rights or other interests or other Securities convertible into any such Securities); (c) any Securities received by CF LLC upon conversion of, in exchange for, as proceeds from the disposition of, as interest on, or as stock dividends or other distributions from, any of the Securities described in clauses (a) or (b) above; or (d) any other investments (including "non-traditional" asset investments such as interest-rate sensitive securities, commodities, futures contracts, derivatives or other hedging contracts or transactions (including hedging for interest rate, currency, commodity and other market and investment risks)) made for the purpose of mitigating the risk of any investment in any Securities described in clauses (a) through (c) above; provided that in no event will the total amount of Capital Contributions used to make any investments described in this clause (d) exceed 10% of the aggregate Capital Contributions made by all Members to CF LLC.

"Sponsor" means BDT & Company, LLC, a Delaware limited liability company.

"Substitute Member" means any Assignee that has been admitted to CF LLC as a Member pursuant to Section 9.6.

"Transfer" means a sale, transfer, assignment, gift, bequest or disposition by any other means, whether for value or no value and whether voluntary or involuntary (including by realization upon any Encumbrance or by operation of law or by judgment, levy, attachment, garnishment, bankruptcy or other legal or equitable proceedings).

The following terms have the respective meanings set forth in the section referenced opposite such term:

<u>Term</u>	Section
"Actions"	6.5.3
"Agreement"	Preamble
"BDT Advisory Business"	Main Fund Agreement
"BDT Funds"	Main Fund Agreement
" <u>CF LLC</u> "	Preamble
"CF LLC Counsel"	11.1
"Closing Date"	Acquisition Agreement
"Commitment Letter"	Co-Investment Fund Agreement
"Co-Investment Fund"	Preamble
" <u>Fair Value</u> "	6.5
"First Common Distribution"	4.2(b)
" <u>Fiscal Year</u> "	11.3
"Fund Advisory Committee"	Main Fund Agreement
"Indemnitee"	6.5.1

"Interim Distributions"	4.2(a)
" <u>Liabilities</u> "	6.5.3
" <u>Main Fund</u> "	Preamble
"Material Adverse Effect"	Main Fund Agreement
" <u>Parallel Fund</u> "	Preamble
"Preferred Distribution"	4.2(d)
" <u>Principals</u> "	Main Fund Agreement
"Purchased Common Shares"	Acquisition Agreement
"Purchased Preferred Shares"	Acquisition Agreement
"Second Common Distribution"	4.2(c)
"Senior Advisory Council"	Main Fund Agreement
" <u>Tax Matters Partner</u> "	8.3
" <u>Tax Proceeding</u> "	8.3

ARTICLE 2 ORGANIZATIONAL MATTERS

- **2.1** <u>Continuation</u>. The Members hereby agree to continue CF LLC as a limited liability company under the Act for the purposes and upon the terms and conditions hereinafter set forth. The rights and liabilities of the Manager and the Members shall be as provided in the Act, except as otherwise expressly provided herein. In the event of any inconsistency between any terms and conditions contained in this Agreement and any non-mandatory provisions of the Act, the terms and conditions contained in this Agreement shall govern. Each Person identified as a Member in CF LLC's books and records on the date hereof has been, or hereby is, admitted to CF LLC as a member of CF LLC at the time that this Agreement or a counterpart hereof is executed by or on behalf of such Person. Upon its execution of this Agreement as of the date hereof, the Manager shall continue as the manager of CF LLC.
- 2.2 Name. The name of CF LLC is "BDT CF Acquisition Vehicle, LLC" The Manager may change the name of CF LLC from time to time, in accordance with applicable law.
- **2.3** <u>Principal Place of Business</u>; <u>Other Places of Business</u>. The principal place of business of CF LLC is located at 401 N. Michigan Avenue, Suite 3100, Chicago, Illinois, 60611, or such other place(s) within or outside the State of Delaware as the Manager may from time to time designate.
- **2.4** <u>Business Purpose</u>. The principal purpose and investment objective of CF LLC is to acquire, hold and dispose of the Co-Investment in accordance with the terms and provisions of this Agreement. CF LLC shall have the power to engage in all activities and transactions which the Manager deems necessary or advisable in connection therewith.
- **2.5** <u>Certificate of Formation; Filings</u>. The Manager has caused to be filed a Certificate of Formation of CF LLC in the Office of the Secretary of State of the State of Delaware as required by the Act. The Manager shall also cause to be made, on behalf of CF LLC, such additional certificates, filings and recordings as the Manager shall deem necessary or advisable.

- **2.6** <u>Designated Agent for Service of Process</u>. So long as is required by the Act, CF LLC shall continuously maintain a registered office and a registered agent for service of process on CF LLC in the State of Delaware.
- **2.7** <u>Term.</u> The term of CF LLC commenced on the date that the Certificate of Formation of CF LLC was initially filed with the Office of the Secretary of State of the State of Delaware, and shall continue until CF LLC is dissolved in accordance with Article 10. Notwithstanding the dissolution of CF LLC, the existence of CF LLC shall continue until termination in accordance with Article 10.

ARTICLE 3 CAPITAL; CAPITAL ACCOUNTS AND MEMBERS

3.1 <u>Commitment to Make Capital Contributions</u>. Each Member, in exchange for its initial Interest, shall make a Capital Contribution in the amount set forth in **Exhibit A** attached hereto and commits to make additional Capital Contributions solely as provided herein.

3.2 Capital Contributions.

- 3.2.1 *Generally*. Except as otherwise required by law or pursuant to this Agreement, no Member shall be required or permitted to make any Capital Contributions. The cumulative Capital Contributions made by each Member at any relevant time during the term of CF LLC shall be set forth in CF LLC's books and records.
 - 3.2.2 Capital Contributions.
 - (a) *Investment*. In order to permit CF LLC to make the Co-Investment, each Member shall make a Capital Contribution in the amount set forth with respect to such member in **Exhibit** A attached hereto. Such Capital Contribution shall be made no later than January ____, 2012 or such other date as may be determined by the Manager. Except as set forth in Section 3.2.2(b) or Section 4.5, no Member will have any obligation to make any additional Capital Contribution. Each such Capital Contribution shall be made in cash.
 - (b) *CF LLC Expenses*. In addition to the foregoing, as and when deemed appropriate from time to time by the Manager as needed to enable CF LLC to: (1) pay CF LLC Expenses; (2) pay any indemnification obligations under Section 6.5; and/or (3) establish adequate reserves for any amounts for which Capital Contributions may be required under clause (1) and clause (2) above (all in accordance with this Agreement), the Manager may require the Members to make Capital Contributions on a *pro rata* basis (in accordance with Percentage Interests) in accordance with procedures to be reasonably determined by the Manager; *provided* that, with respect to the CF LLC Expenses described in clause (1) above, the Manager shall first cause such expenses to be satisfied out of the assets of CF LLC to the extent the Manager determines in good faith that there are sufficient such assets available therefor without the need to liquidate any Securities; and *provided*, *further* that, with respect to the indemnification obligations described in clause (2) above, any Capital Contributions to be made by the Members shall be limited to the return of prior distributions in accordance with the procedures set forth in Section 6.5.4.

- 3.2.3 *No Third-Party Beneficiaries*. Except to the extent provided in the Commitment Letters, unless separately agreed to in writing by the Manager, the provisions of this Section 3.2 are intended solely to benefit CF LLC, the Members, and the Fund LPs, and, to the fullest extent permitted by applicable law, will not confer any benefit upon any creditor of CF LLC (and no such creditor will be a third-party beneficiary of this Agreement), and no Member will have any duty or obligation to any such creditor to make any Capital Contributions or to cause the Manager to deliver any capital call notice.
- 3.3 Failure to Contribute. Notwithstanding anything to the contrary contained in this Agreement, if any Member fails to timely contribute all or any portion of its required Capital Contributions as a result of the failure of any "Defaulting Partner" (as defined in the Co-Investment Fund Agreement, the Main Fund Agreement, or the Parallel Fund Agreement) to make any required capital contributions in accordance with the provisions of the Co-Investment Fund Agreement, the Main Fund Agreement, or the Parallel Fund Agreement, then the Manager may take such actions as the Manager determines to be necessary or appropriate to facilitate the implementation of the provisions of Section 3.3 of the Co-Investment Fund Agreement, the Main Fund Agreement, or the Parallel Fund Agreement, as applicable, with respect to such Defaulting Partner's failure.
 - 3.4 Capital Accounts. A Capital Account shall be established and maintained for each Member in accordance with the terms of this Agreement.
- **3.5** <u>Limited Liability of the Members</u>. Notwithstanding anything to the contrary contained in this Agreement and except as otherwise required by any non-waivable provision of applicable law, the liability of a Member (in its capacity as such) for any liabilities, obligations, expenses or losses of CF LLC in no event shall exceed, in the aggregate: (a) its obligation to make Capital Contributions pursuant to the terms of this Agreement; plus (b) its share of undistributed assets and profits of CF LLC; plus (c) without duplication, its obligation to make any payments expressly required to be made by this Agreement.
- **3.6** <u>Admission of Additional Members</u>. Except in connection with the admission of any Substitute Member pursuant to Article 9, after the date hereof, one or more Persons may be admitted as additional Members of CF LLC only with the consent of the Manager and a Majority in Interest and the execution of a counterpart signature page to this Agreement by or on behalf of such newly admitted Members.
- **3.7** <u>Member Capital</u>. Except as otherwise provided in this Agreement: (a) no Member shall demand or be entitled to receive a return of or interest on its Capital Contributions or Capital Account, (b) no Member shall withdraw any portion of its Capital Contributions or receive any distributions from CF LLC as a return of capital on account of such Capital Contributions, and (c) CF LLC shall not redeem the Interest of any Member.

3.8 Member Loans. No Member shall be required to make any loans or otherwise lend any funds to CF LLC, and no Member shall be permitted to do so except with the consent of the Manager. The Manager shall be permitted to make loans to, act as surety or endorser for, assume one or more specific obligations of, provide collateral for, or enter into other similar credit, guarantee, financing or refinancing arrangements with, CF LLC for any purpose (including in connection with any investment), to the extent the Manager determines that such loans are necessary or advisable for the business of CF LLC, *provided* that any loans made to CF LLC by the Manager or any of its Affiliates shall be on terms no less favorable to CF LLC than it could otherwise obtain from third parties under similar circumstances.

ARTICLE 4 DISTRIBUTIONS

4.1 Distributions Generally.

- 4.1.1 Except as otherwise expressly provided in this Agreement, no Member may withdraw capital from CF LLC or receive any distribution or return of its Capital Contributions.
- 4.1.2 Each distribution made by CF LLC shall be made to the Persons who, according to the books and records of CF LLC, were the holders of record of Membership Interests on the date determined by the Manager as of which the Members are entitled to such distribution, and shall be made to the Members on a *pro rata* basis in accordance with the relative Percentage Interests of the Members entitled to receive such distribution as of such date. Distributions of cash will be made to each Member in U.S. dollars by internal account transfers or by wire transfer of immediately available funds to the account specified in **Exhibit A** attached hereto or as otherwise specified in writing by such Member to the Manager. Distributions in kind shall be made in accordance with Section 4.4.
- 4.1.3 Fees and reimbursements received by the Manager and/or any Affiliate(s) thereof (including pursuant to Article 6) are not, and shall not be deemed to be, distributions.
- 4.1.4 Notwithstanding any contrary provision in this Agreement, neither CF LLC, nor the Manager on behalf of CF LLC, shall be required to make a distribution to any Member pursuant to this Agreement if such distribution would violate any non-waivable provision of the Act or other applicable law.
- 4.1.5 Notwithstanding any contrary provision in this Article 4, prior to the dissolution of CF LLC, the Manager may cause CF LLC to retain and not distribute to the Members such portion of any Available Assets to be used or reserved for the payment of CF LLC Expenses or for any other purpose permissible under this Agreement.
- **4.2** <u>Timing and Manner of Distributions</u>. Except as set forth in Sections 4.3, 4.5, and 9.9, and in Article 10, distributions shall be made to the Members in the following manner and at the following specified times:
 - (a) *Interim Distributions of Co-Investment Proceeds*. Promptly following CF LLC's receipt of any Co-Investment Proceeds, CF LLC shall distribute (any such distribution an "Interim Distribution") all Available Assets received in connection therewith to the Members in accordance with Section 4.1.2.

- (b) *First Common Distribution*. Without limitation of Section 4.2(a), by not later than nine (9) months following the Closing Date, CF LLC shall distribute (the "<u>First Common Distribution</u>") at least fifty percent (50%) of the Common Investment Assets (determined by reference to the number of shares of Common Stock (as defined in the Acquisition Agreement) included in the Common Investment Assets or such other manner of calculation as the Manager shall elect in its reasonable discretion) to the Members in accordance with Section 4.1.2, to the extent of Available Assets.
- (c) *Second Common Distribution*. Without limitation of Section 4.2(a) and Section 4.2(b), by not later than fifteen (15) months following the Closing Date, CF LLC shall distribute (the "Second Common Distribution") one hundred percent (100%) of the Common Investment Assets to the Members in accordance with Section 4.1.2, to the extent of Available Assets.
- (d) *Preferred Distribution*. Without limitation of Section 4.2(a), by not later than five (5) years following the Closing Date, CF LLC shall distribute (the "<u>Preferred Distribution</u>") one hundred percent (100%) of the Preferred Investment Assets to the Members in accordance with Section 4.1.2, to the extent of Available Assets. Promptly following the payment in full of the Preferred Distribution, CF LLC shall commence liquidation and winding up in accordance with Section 10.2(g).
- (e) *Other Distributions*. Any distributions other than Interim Distributions, the First Common Distribution, the Second Common Distribution, and the Preferred Distribution, may be made in accordance with Section 4.1.2 out of Available Assets from time solely as determined by the Manager.
- **4.3** <u>Distributions upon Liquidation</u>. Distributions made in conjunction with the final liquidation of CF LLC shall be applied or distributed as provided in Article 10 hereof.
- **4.4** <u>Distributions In-Kind</u>. No Member shall demand, nor, except as set forth in this Section 4.4, shall any Member receive, property other than cash. Without the consent of a Majority in Interest, the Manager shall not make distributions in kind other than in the form of Marketable Securities that do not exceed applicable volume limitations (including any volume restrictions pertaining to Rule 144 of the Securities Act of 1933, as amended) and which are transferable free and clear of any lock-ups or other legal or contractual restrictions on transferability as of the date of such distribution, except: (a) in connection with the liquidation of CF LLC in accordance with Section 10.4; or (b) in connection with the First Common Distribution, the Second Common Distribution, and the Preferred Distribution. The Manager may require that, as a condition to any Member receiving any distribution in kind, such Member will make customary representations, warranties and covenants in connection therewith as the Manager deems necessary or appropriate. Any in-kind distributions shall be made in such a fashion as to ensure that the Fair Value is distributed and allocated in accordance with Sections 6.4 and 10.4, this Article 4 and Article 5 (as applicable) and that each Member receives not greater than its proportionate share (as determined in accordance with this Article 4, and Section 10.4) of such in-kind distributions (except as otherwise provided in this Agreement).

4.5 Taxes; Withholding. CF LLC may withhold distributions or portions thereof if it is required to do so by any applicable rule, regulation, or law. Each Member hereby authorizes CF LLC to withhold from or pay on behalf of or with respect to such Member any amount of federal, state, local or foreign taxes that the Manager determines in good faith that CF LLC is required to withhold or pay with respect to any amount distributable or allocable to such Member pursuant to this Agreement. The amount of any such taxes paid by or withheld from receipts of CF LLC from an investment (or any flow-through vehicle in which it invests) will be allocated to the Members to which such withholding relates. Any amounts so withheld or paid on behalf of or with respect to a Member pursuant to this Section 4.5 due to the status, action, or inaction of a Member shall be deemed to have been distributed to such Member (except to the extent repaid to CF LLC by such Member in accordance with this Section 4.5). To the extent that the cumulative amount of such withholding for any period exceeds the distributions to which such Member is entitled for such period, the amount of such excess shall be considered a loan from CF LLC to such Member, with interest at 6% until discharged by such Member by repayment, which may, at the option of the Manager, be satisfied (a) out of distributions to which such Member would otherwise be subsequently entitled, or (b) by the immediate payment in cash to CF LLC of such excess amount. The Manager, on behalf of CF LLC, shall be entitled to take any other action it determines to be necessary or appropriate in connection with any obligation or possible obligation to impose withholding pursuant to any applicable rule, regulation, or law or to pay any tax with respect to a Member. Each Member hereby unconditionally and irrevocably grants to CF LLC as security interest in such Member's obligations as CF LLC shall request in order to perfect or enforce the security interest created hereunder. Each Member's obligat

4.6 Tax Law Changes. If the Manager determines in good faith that there has been a change in international, U.S. federal, state or local tax law that is adverse to the Manager (or to the direct or indirect owners of the Manager), then the Manager may, with the approval of at least 66-2/3% in Interest, cause a restructuring of CF LLC and/or of the Co-Investment in a manner which the Manager intends to mitigate the adverse impact of such change on the Manager and its direct and indirect owners while preserving the overall intended relative economic arrangements among the Members contemplated by this Agreement; provided, however, that no such restructuring which would adversely affect any Member or Fund LP shall be made without the consent of such Member or such Fund LP, as applicable.

ARTICLE 5 ALLOCATIONS OF NET INCOME AND NET LOSS

5.1 <u>Timing and Amount of Allocations of Net Income and Net Loss</u>. Net Income and Net Loss shall be determined and allocated for each Fiscal Year as of the end of each such year, at any time the Gross Asset Value of any CF LLC property is adjusted, and more often as required, or, in the Manager's good faith discretion, as permitted hereby or by the Code. Except as otherwise provided herein, subject to the other provisions of this Agreement, an allocation to a Member of a share of Net Income or Net Loss shall be treated as an allocation of the same share of each item of income, gain, loss or deduction that is taken into account in computing Net Income or Net Loss.

- 5.2 General Allocation Provisions. Subject to Section 5.3, Net Income, Net Loss and, to the extent necessary, individual items of income, gain, loss, credit and deduction, for any Fiscal Year shall be allocated among the Members in a manner such that the Adjusted Capital Account (as determined without regard to clause (b) of the definition of "Adjusted Capital Account") of each Member, immediately after making such allocation is, as nearly as possible, equal (proportionately) to the distributions that would be made to such Member pursuant to this Agreement if CF LLC were dissolved, its affairs wound up and its assets sold for cash equal to their Gross Asset Values, all CF LLC liabilities were satisfied (limited with respect to each nonrecourse liability (as defined in Reg. §§ 1.704-2(b)(3) and 1.752-1(a)(2)) to the Gross Asset Value of the asset securing such liability), and the net assets of CF LLC were distributed in accordance with Section 4.1.2 immediately after making such allocation.
- **5.3** <u>Additional Allocation Provisions</u>. Notwithstanding the foregoing provisions of this Article 5, the additional allocation provisions set forth on Annex I hereto shall apply.

ARTICLE 6 OPERATIONS

6.1 Authority of the Manager.

- 6.1.1 *General*. The management, control, operation and policy of CF LLC shall be vested exclusively in the Manager (including its duly appointed agents), which shall have the power by itself (or through such agents), and shall be authorized and empowered on behalf and in the name of CF LLC, to carry out any and all of the objects and purposes of CF LLC and to perform all acts (including the payment of CF LLC obligations) and enter into and perform all contracts and other undertakings, consistent with the provisions of this Agreement, that it may deem necessary or advisable or incidental thereto. The Manager may appoint officers of CF LLC, who shall have such titles and duties, as the Manager shall determine from time to time.
- 6.1.2 *Limitations on the Manager*. Notwithstanding anything in this Agreement to the contrary, the Manager and its Affiliates shall not take any of the following actions without the consent of a Majority in Interest:
 - (a) amend or modify the certificate of formation of CF LLC; or
 - (b) elect to cause CF LLC not to be treated as a partnership for U.S. federal income tax purposes.
- 6.1.3 *Appointment of Officers*. Effective on the date hereof, each of the following officers is hereby appointed in the capacity set forth opposite his name until such time as his successor is duly appointed and qualified or his earlier displacement by the Manager, resignation, removal or otherwise:

Nam	Title
Byron D. Trott	President and Chief Investment Officer
William R. Bush	Vice President, General Counsel and Secretary
San W. Orr, III	Vice President and Treasurer

Each of the foregoing officers, and any other officers appointed by the Manager from time to time pursuant to Section 6.1.1, shall have all power and authority that may be delegated to such officer by the Manager to operate and direct the business and operations of CF LLC, including, without limitation, the power and authority to prepare, execute, and deliver, acknowledge, attest, file, and record, as applicable, any and all agreements, instruments, certificates and other documents on behalf of CF LLC.

6.2 <u>No Member Management</u>. No Member shall have any right or authority to act for or on behalf of or to bind CF LLC. No Member shall participate in the management of CF LLC or have any control over CF LLC business, except as otherwise expressly provided herein.

6.3 Other Activities.

- 6.3.1 General Prohibitions.
- (a) Except in connection with the Placement Agent Agreement, the Engagement Letter, the Acquisition Agreement (as in existence as of the date hereof), and the Co-Investment Fund Agreement and as otherwise permitted under or expressly set forth in this Agreement, none of the Manager or its Affiliates shall engage in any transaction with CF LLC except on an arm's-length basis and on terms no less favorable to CF LLC than would be available from third parties under similar circumstances; *provided* that, the Manager shall provide to each Member notice of any such transaction no later than the later of thirty days after the occurrence of such transaction, or the end of the calendar quarter during which such transaction occurred, which notice shall include the terms of such transaction and copies of all agreements, if any, executed in connection with such transaction.
- (b) None of the Manager or its Affiliates (other than the Co-Investment Company) shall purchase any Securities or assets from, or sell any Securities or assets to, CF LLC.
 - (c) The Manager shall not take any action that would reasonably be expected to result in the loss of limited liability for any Member.
 - 6.3.2 *Permitted Activities*. Except as otherwise expressly provided herein:
- (a) Affiliates of the Manager and the Sponsor, and their respective partners, members, directors, officers, shareholders, and employees, and the Principals, shall have the benefit of Sections 6.3.2(a) and (b) of the Main Fund Agreement as if such provisions were set forth herein, *mutatis mutandis*.
- (b) Affiliates of the Manager and their respective partners, members, directors, officers, shareholders, and employees, shall have no obligation to purchase or sell for CF LLC any investment which such Persons may purchase or sell, or recommend for purchase or sale, for its or their own accounts, or for the account of any other Person (except as otherwise required by this Agreement).

- (c) Neither CF LLC nor any Member shall have any rights of first refusal, co-investment or other rights in respect of the investments of other accounts or the conduct of any other business (including the BDT Advisory Business) or from any transaction in Securities effected by the Manager, any of its Affiliates, their respective partners, directors, officers, shareholders or employees, or the Principals for any account other than that of CF LLC or in any fees, profits or other income earned or otherwise derived therefrom.
- (d) Without limiting the generality of the foregoing, and notwithstanding any other provision herein to the contrary, the Members expressly acknowledge and agree that BDT & Company, LLC is permitted to enter into, and perform its obligations under, the Placement Agent Agreement and the Engagement Letter, and, shall, except as set forth in the Main Fund Agreement and in the Parallel Fund Agreement, or as otherwise agreed in writing between any Member (and/or such Member's beneficial owners) and BDT & Company, LLC, be entitled to receive and retain all fees earned thereunder without any obligation to account for the same to CF LLC, to the Members, or to any of their respective beneficial owners.
- 6.3.3 *BDT Advisory Business*. Notwithstanding any contrary provision contained in this Agreement, Affiliates of the Manager and their respective partners, members, directors, officers, shareholders, and employees, shall be free to conduct all aspects of the BDT Advisory Business, including the receipt of advisory and similar fees in connection therewith. Neither CF LLC nor any other Member shall have any right in such BDT Advisory Business or any economic interest therein.
- **6.4** <u>Valuation</u>. The calculation of the fair value (the "<u>Fair Value</u>") of all CF LLC assets shall be made in good faith by the Manager consistent with the manner in which the valuation of assets is established for the Main Fund under Section 6.6 of the Main Fund Agreement. For all purposes of this Agreement, all valuations made pursuant to this Section 6.4 shall be final, conclusive and binding on CF LLC, all Members, and the respective successors and assigns of the foregoing.

6.5 Liability; Indemnification.

6.5.1 No present or former Sponsor, Manager, Principal, Tax Matters Partner, officer of CF LLC, or Affiliates of any of the foregoing, or any of their respective present or former partners, shareholders, members, directors, officers, employees and agents, and any person who serves as a member of the Fund Advisory Committee and/or the Senior Advisory Council (in each case, an "Indemnitee" and collectively, the "Indemnitees"), shall be liable to CF LLC or to any Member for any losses sustained or liabilities incurred as a result of any act or omission taken or suffered by such Indemnitee(s) unless a court of competent jurisdiction determines in a final, nonappealable judgment that (a) such Indemnitee(s) did not act in good faith, or (b) the conduct of such Indemnitee(s) constituted Malfeasance (provided that in the case of an Indemnitee that is a member of the Fund Advisory Committee and/or the Senior Advisory Council, or the entity that such a Fund Advisory Committee member represents, the term "Malfeasance" as used in this Section 6.5 shall be limited to any act or omission that constitutes fraud or willful misconduct).

6.5.2 The Indemnitees shall not be liable to CF LLC or to any Member for any action taken by any other Member, nor shall any such Indemnitee be liable to CF LLC or to any Member for any action of any agent or employee of CF LLC or the Manager (in the absence of Malfeasance by such Indemnitee in connection with such action and/or the selection, retention and/or supervision of such agent or employee).

6.5.3 CF LLC shall indemnify and hold harmless, and the Members shall release, each Indemnitee to the fullest extent permitted by law from and against any and all losses, claims, demands, costs, damages, liabilities, expenses of any nature (including attorneys' fees and disbursements), judgments, fines, settlements and other amounts, of any nature whatever, known or unknown, liquid or illiquid (collectively, "Liabilities") arising from any and all claims, demands, actions, suits or proceedings, whether civil, criminal, administrative or investigative (collectively, "Actions"), in which the Indemnitee may be involved, or threatened to be involved as a party or otherwise, relating to the performance or nonperformance of any act concerning the activities of CF LLC and/or the Co-Investment, unless a court of competent jurisdiction determines in a final, nonappealable judgment that (a) the Indemnitee did not act in good faith, or (b) the Indemnitee's conduct constituted Malfeasance; provided that CF LLC's responsibility for any liabilities arising in respect of the investments in the Co-Investment Company made collectively by CF LLC, the Co-Investment Fund, the Main Fund, and the Parallel Fund (and/or their respective Affiliates) shall be limited to CF LLC's equitable share thereof (as determined by the Manager in its good faith discretion). The termination of an action, suit or proceeding by judgment, order, settlement, or upon a plea of nolo contendere or its equivalent, shall not, in and of itself, create a presumption or otherwise constitute evidence that the Indemnitee acted in the manner specified in clause (a) or (b) above. Notwithstanding the foregoing, the Indemnitees shall not be entitled to indemnification hereunder with respect to disputes solely among and between the Manager, the Sponsor, the Tax Matters Partner and any of their respective Affiliates.

6.5.4(a) Any indemnification obligations of CF LLC arising under this Section 6.5 may, but shall not be required to, be satisfied first out of any CF LLC assets; *provided* that the Manager shall use commercially reasonable efforts to obtain recovery from any insurance policies then in effect covering the applicable Indemnitee prior to CF LLC advancing indemnification payments to such Indemnitee. In addition, in accordance with Section 3.2 and subject to the limitations contained in this Section 6.5.4, the Manager may require the Members to return distributions previously received (which shall be treated as additional Capital Contributions), whether or not CF LLC has been dissolved, to satisfy all or any portion of the indemnification obligations of CF LLC pursuant to this Section 6.5 in connection with any Liabilities arising from any and all Actions of which the Manager receives notice or otherwise becomes aware at any time during the existence, or prior to the date that is three years following the termination, of CF LLC.

- (b) The Manager shall cause any distributions to be returned by the Members under this Section 6.5.4 to be so returned in the reverse order of priority in which such distributions were previously made pursuant to Sections 4.2 and 10.4(c), with the intention being that the Members' relative aggregate net distributions (after giving effect to any such returns) shall remain consistent with the distribution priorities set forth in Articles 4 and 10. The Manager is hereby authorized to adjust the relative amounts to be returned hereunder as needed to effect such intention to the maximum extent possible.
- (c) The Manager shall give the Members timely written notice of any such Actions of which it becomes aware. Upon receipt of a capital call notice from the Manager in respect of distributions required to be returned in accordance with this Section 6.5.4 and otherwise in accordance with Section 3.2, each Member shall make a Capital Contribution to CF LLC in respect of its share of any such indemnification required to be made pursuant to this Section 6.5.4.
- (d) Notwithstanding anything in this Section 6.5 to the contrary: (i) no Member shall be required to return any particular distribution after the third anniversary of the date of its receipt of such distribution; *provided* that if at or prior to the end of such period, there are any Actions then pending or any other Liability (whether contingent or otherwise) then outstanding, the Manager shall so notify the Members at such time (which notice shall include a brief description of each such Action or of such Liabilities) and the obligation of the Members to return such distribution for the purpose of meeting any obligations of CF LLC relating thereto shall survive with respect to each such Action and Liability set forth in such notice (or any related Action or Liability based upon the same or a similar claim) until the date that such Action or Liability is ultimately resolved and satisfied; and *provided*, *further*, that the provisions of this Section 6.5.4(d) shall not affect the obligations of the Members under Sections 18-607 and 18-804 of the Act or other applicable law; and (ii) no Member shall be individually obligated to return distributions pursuant to this Section 6.5.4 in an aggregate amount exceeding 25% of the greater of: (A) its original Capital Contribution pursuant to Section 3.2.2(a) and (B) the distributions received by such Member from CF LLC, *provided* that, in no event shall the aggregate amount of a Member's obligations therefor exceed the aggregate distributions so received by such Member.
- 6.5.5 Expenses incurred by an Indemnitee in defending any Action subject to this Section 6.5 may, in the Manager's sole discretion, be advanced by CF LLC prior to the final disposition of such Action: (a) if the Manager reasonably believes that such Indemnitee may be entitled to indemnification in respect of such Action; and (b) upon receipt by CF LLC of an enforceable written commitment by or on behalf of the Indemnitee to repay such amount (i) if it shall be determined by any final, non-appealable judgment entered by any court of competent jurisdiction that such Indemnitee acted in the manner specified in clause (a) or (b) of Section, 6.5.3 and/or (ii) if, and to the extent that, such Indemnitee receives indemnification payments from the Co-Investment Company, the Co-Investment Fund, the Manager, the Main Fund, or the Parallel Fund for the same Liabilities (or the same portion thereof).
- 6.5.6 The provisions of this Section 6.5 are for the sole benefit of the Indemnitees, and such Indemnitees shall have an independent right of enforcement in respect of Section 6.5 as intended third-party beneficiaries hereunder. The provisions of this Section 6.5 shall not be deemed to create any rights for the benefit of any other Person (except as provided in the immediately preceding sentence).

6.5.7 To the full extent permitted by law, the Manager is hereby authorized on behalf of CF LLC to indemnify, hold harmless and release any agents and/or advisors of CF LLC, the Manager and their respective Affiliates, to the same extent provided with respect to the Indemnitees in this Article 6.

6.5.8 Each Indemnitee shall use its commercially reasonable efforts to pursue any insurance or indemnity claims it may have against third parties (including pursuant to any insurance policy acquired by CF LLC, the Co-Investment Fund, the Main Fund, the Parallel Fund, or the Co-Investment Company in connection therewith) with respect to any liabilities relating to any Action, *provided* that, subject to Section 6.5.9, no obligation under this Section 6.5.8 shall delay the availability of the advances otherwise available pursuant to Section 6.5.5.

6.5.9 The indemnification obligation of CF LLC to an Indemnitee with respect to any Liabilities shall be reduced by any indemnification payments actually received by such Indemnitee from the Co-Investment Fund, the Main Fund, any Parallel Fund or the Co-Investment Company with respect to the same Liabilities. Solely for purposes of clarification, and without expanding the scope of indemnification pursuant to this Section 6.5, the Members and the Manager intend that, to the maximum extent permitted by law, as between the Co-Investment Company, on the one hand, and CF LLC, the Co-Investment Fund, the Main Fund, and the Parallel Fund, on the other hand, this Section 6.5 shall be interpreted to reflect an ordering of liability for potentially overlapping or duplicative indemnification payments, with the Co-Investment Company having primary liability, and CF LLC, the Main Fund, the Parallel Fund, and the Co-Investment Fund having only secondary liability, proportionate to such funds' relative direct or indirect ownership of the Co-Investment Company. The possibility that an Indemnitee may receive indemnification payments from the Main Fund, the Parallel Fund, and the Co-Investment Company shall not restrict CF LLC from making payments under this Section 6.5 to an Indemnitee that is otherwise eligible for such payments under Section 6.5.3 and shall not delay the availability of advances otherwise available pursuant to Section 6.5.5 to the extent no such advances are available from the Co-Investment Fund, the Main Fund, the Parallel Fund, the Co-Investment Company and/or other third parties, but such payments or advances by CF LLC are not intended to relieve the Co-Investment Company, the Main Fund, any Parallel Fund, and the Co-Investment Company from any liability that it would otherwise have to make indemnification payments to such Indemnitee and, if an Indemnitee that has received indemnification payments or advances from CF LLC actually receives duplicative indemnification payments or advances from the Co-Investment Fund, the Main Fund, the Parallel Fund and the Co-Investment Company for the same Liabilities, such Indemnitee shall repay CF LLC to the extent of such duplicative payments or advances. If, notwithstanding the intention of this Section 6.5, the Co-Investment Fund, the Main Fund, the Parallel Fund, and the Co-Investment Company's obligation to make indemnification payments or advances to an Indemnitee is relieved or reduced under applicable law as a result of payments or advances made by CF LLC pursuant to this Section 6.5, CF LLC shall have, to the maximum extent permitted by law, a right of subrogation against (or contribution from) each of the Co-Investment Fund, the Main Fund, the Parallel Fund and the Co-Investment Company for amounts paid or advanced by CF LLC to an Indemnitee that relieved or reduced the obligation of the Co-Investment Fund, the Main Fund, the Parallel Fund, and the Co-Investment Company to such Indemnitee.

6.5.10 The Manager and its Affiliates shall at all times act in a manner that is consistent with its implied contractual covenant of good faith and fair dealing. So long as such Persons act in a manner consistent with the implied contractual covenant of good faith and fair dealing and with the express provisions of this Agreement, such Persons shall not be in breach of any duties (including fiduciary duties) in respect of CF LLC and/or any Member otherwise applicable at law or in equity. The provisions of this Agreement, to the extent that they expand, restrict or eliminate the duties and liabilities of such Persons otherwise existing at law or in equity, are agreed by the Manager and the Members to replace fully and completely such other duties and liabilities of such Persons. Subject to the foregoing but notwithstanding any other provision of this Agreement or otherwise applicable provision of law or equity, whenever in this Agreement the Manager is permitted or required to make a decision or take an action (a) in its "sole discretion" or "discretion" or under a similar grant of authority or latitude, in making such decisions, the Manager shall be entitled to take into account its own interests as well as the interests of the Members (and the investors of the BDT Funds) as a whole or (b) in its "good faith" or under another express standard, the Manager shall act under such express standard and shall not be subject to any other or different standards.

6.5.11 The Manager may consult with legal counsel and accountants (and other similar experts) and any act or omission suffered or taken by the Manager on behalf of CF LLC in furtherance of the interests of CF LLC in good faith in reliance upon and in accordance with the advice of such counsel or accountants (or other similar experts) will be full justification for any such act or omission, and the Manager will be fully protected (and shall not be liable to CF LLC or any Member) in so acting or omitting to act, so long as such counsel or accountants (or other similar experts) were selected in good faith. In addition, if the Manager consults with the Fund Advisory Committee with respect to a matter giving rise to a conflict of interest, and if the Fund Advisory Committee waives such conflict of interest or the Manager acts in a manner, or pursuant to standards or procedures, approved by the Fund Advisory Committee with respect to such conflict of interest, then none of the Manager, the Principals, or any of their respective Affiliates shall have any liability to CF LLC or any Member for such actions in respect of such matter taken in good faith by them, including actions in the pursuit of their own interests, and such actions shall not constitute a breach of this Agreement or any other agreement contemplated herein or of any duty or obligation of such Person at law or in equity or otherwise.

6.6 Fees and Expenses. The Manager will bear and be charged with all Manager Expenses. The costs and expenses of CF LLC and (to the extent fairly allocable to CF LLC) of the Manager which constitute CF LLC Expenses will be borne by and charged to CF LLC. The Manager shall be entitled to reimbursement from CF LLC for any CF LLC Expenses paid by the Manager subject to the terms of this Agreement.

ARTICLE 7
INVESTMENT LIMITATIONS

7.1 Member Investment Limitations. CF LLC shall conduct all of its business, including, without limitation, the acquisition, holding, and disposition of the Co-Investment and any other investments, in compliance with the provisions of Article 7 of each of the Co-Investment Fund Agreement, the Main Fund Agreement, and the Parallel Fund Agreement. Each Member acknowledges and agrees that none of the incurrence of indebtedness by the Co-Investment Fund pursuant to Section 3.8 of this Agreement or by the Co-Investment Fund pursuant to Section 3.8 of the Co-Investment Fund Agreement, the operation of Section 6.8.2(c) of each of the Main Fund Agreement and the Parallel Fund Agreement, or any other provisions or arrangements specifically contemplated by this Agreement or the Co-Investment Fund Agreement shall be deemed to violate this Section 7.1.

ARTICLE 8 BOOKS AND RECORDS; ACCOUNTING; TAX ELECTIONS

- **8.1** Books and Records. The Manager shall keep or cause to be kept, at the principal place of business of CF LLC, or at such other location as the Manager shall reasonably deem appropriate (with notice thereof to the Members), full and proper ledgers, other books of account, and records of all receipts and disbursements, other financial activities, and the internal affairs of CF LLC for the current and prior Fiscal Years. The books of CF LLC shall be maintained, for financial reporting purposes, on an accrual basis, in accordance with United States generally accepted accounting principles, consistently applied. Except as otherwise expressly set forth herein, all decisions as to accounting matters shall be made by the Manager in its good faith discretion.
- **8.2** <u>Delivery of Records; Inspection; Reports to Fund LPs.</u> The Manager shall comply, and shall cause CF LLC to comply, with the requirements of Sections 8.1, 8.2, and 8.3 of the Co-Investment Fund Agreement, the Main Fund Agreement, and the Parallel Fund Agreement, including, without limitation, by providing access to the books and records of CF LLC and furnishing financial reports to the Fund LPs, in each case to the extent required by the Co-Investment Fund Agreement, the Main Fund Agreement, or the Parallel Fund Agreement, as applicable, and subject to all terms, provisions, limitations, restrictions, and conditions set forth therein (including, without limitation, provisions as to confidentiality).
- **8.3** <u>CF LLC Tax Elections; Tax Controversies.</u> The Manager shall have the right in its good faith discretion to make all elections for federal income and all other tax purposes (including, but not limited to, the election provided for in Section 754 of the Code); *provided, however*, prior to making any material tax election or decision, the Manager will consult with any Member or Fund LP that would reasonably be expected to be subject to a disproportionately adverse consequence as a result of such tax election or decision. The Main Fund is hereby designated as the "<u>Tax Matters Partner</u>" (as defined in Section 6231(a)(7) of the Code) and in such capacity shall represent CF LLC in any disputes, controversies or proceedings with the Internal Revenue Service. Within five (5) Business Days after the receipt thereof, the Tax Matters Partner shall provide the Members and Fund LPs with copies of all material notices, reports, demands, or other written communications received by CF LLC or the Tax Matters Partner from, or sent by CF LLC or the Tax Matters Partner to, any governmental authority in connection with any audit, examination, or judicial or administrative proceeding with respect to taxes imposed by any jurisdiction (any such audit, examination, or judicial or administrative proceeding"). The Members and Fund LPs shall have the right (but shall not be required) to participate (at their sole cost and expense) in any Tax Proceeding, and the Tax

Matters Partner shall keep the Members and Fund LPs informed, on a timely basis, of every material aspect of any Tax Proceeding and, in addition, shall consult with the Members and Fund LPs regarding every material aspect of any Tax Proceeding.

ARTICLE 9 <u>INTERESTS; TRANSFERS AND</u> ENCUMBRANCES OF INTERESTS

- **9.1** Member Transfers. To the fullest extent permitted by law, and except as provided in the following sentence and in Section 9.9, no Member or Assignee thereof may Transfer all or any portion of its Interest (or beneficial interest therein), directly or indirectly, without the prior written consent of the Manager, which consent may be given or withheld, or made subject to such conditions as are determined by the Manager, in the Manager's sole discretion. Notwithstanding the immediately preceding sentence, but subject to Sections 9.4 and 9.7, any Member or Assignee may Transfer all or any portion of its Interest (or beneficial interest therein), without the consent of the Manager to: (a) any Affiliate (for so long as such Affiliate remains an Affiliate of the transferor); provided that if any Member (or Assignee thereof) Transfers all or any portion of its Interest (or beneficial interest therein) to an Affiliate and such Affiliate ceases to be an Affiliate of the transferor at any time thereafter, then such transferor and such transferee shall be in material breach of this Agreement with respect to all Interests held by either of them in CF LLC; (b) any other Member; and/or (c) any member of such Member's Immediate Family. To the fullest extent permitted by law, any purported Transfer which is not in accordance with, or subsequently violates, this Agreement shall be null and void.
- **9.2** Encumbrances. No Member or Assignee may create an Encumbrance with respect to all or any portion of its Interest (or any beneficial interest therein) unless the Manager consents in writing thereto, which consent may be given or withheld, or made subject to such conditions as are determined by the Manager, in the Manager's sole discretion. To the fullest extent permitted by law, any purported Encumbrance which is not in accordance with this Agreement shall be null and void.
- **9.3** <u>Further Restrictions</u>. Notwithstanding any contrary provision in this Agreement, unless waived in writing by the Manager in its sole discretion, any otherwise permitted Transfer or Encumbrance shall, to the fullest extent permitted by law, be null and void if:
 - (a) such Transfer or Encumbrance could cause a termination of CF LLC for federal or state, if applicable, income tax purposes;
 - (b) such Transfer or Encumbrance could cause CF LLC to cease to be classified as a partnership for federal or state income tax purposes;
 - (c) such Transfer or Encumbrance could require the registration of such Transferred Interest or other Interests pursuant to any applicable federal or state securities laws;
 - (d) such Transfer or Encumbrance could cause CF LLC to become a "publicly traded partnership," as such term is defined in Sections 469(k)(2) or 7704(b) of the Code;

- (e) such Transfer or Encumbrance could cause CF LLC to fail to meet the "lack of actual trading" safe harbor or any other safe harbor from treatment as a "publicly traded partnership" selected by the Manager, as described in Reg. § 1.7704-1;
- (f) such Transfer or Encumbrance could involve Interests being traded on an "established securities market" or a "secondary market or the substantial equivalent thereof" as those terms are defined in Reg. § 1.7704-1 (in addition, such Transfers shall not be "recognized" (as that term is defined in Reg. § 1.7704-1(d)(2)) by CF LLC);
- (g) such Transfer or Encumbrance could subject CF LLC, the Co-Investment Fund, the Main Fund, the Parallel Fund, or the Manager to regulation under the Investment Company Act, the Investment Advisers Act or ERISA;
- (h) such Transfer or Encumbrance could constitute a non-exempt prohibited transaction under ERISA or could cause any portion of the assets of CF LLC to constitute "plan assets" under ERISA or the Code or to become subject to the provisions of ERISA;
 - (i) such Transfer or Encumbrance could result in a violation of any applicable laws, regulations or administrative orders;
 - (j) such Transfer or Encumbrance is made to any Person who could lack the legal right, power or capacity to own such Interest;
 - (k) such Transfer or Encumbrance could cause a Material Adverse Effect; or
- (l) CF LLC does not receive written instruments (including investor representations, copies of any instruments of Transfer, written opinions of counsel, and such Assignee's consent to be bound by this Agreement as an Assignee) that are in form and substance satisfactory to the Manager (as determined in the Manager's sole discretion).

In its sole discretion, the Manager may condition its consent to any Transfer or Encumbrance upon receipt of a written opinion of counsel of the Manager or Assignee (which opinion and counsel shall be reasonably satisfactory to the Manager) that such Transfer or Encumbrance will not result in the occurrence of any of the events described in clauses (a) through (k) above. Each assigning Member agrees that it will pay all out-of-pocket expenses, including attorneys' fees, incurred by CF LLC in connection with an assignment or transfer of a Interest by such Member, except to the extent that the Assignee thereof agrees to bear such expenses.

9.4 <u>Rights of Assignees</u>. Subject to Section 9.6, the transferee of any permitted Transfer pursuant to this Article 9 shall be an Assignee only, and only shall receive, to the extent Transferred, the distributions and allocations of income, gain, loss, deduction, credit, or similar items to which the Member that Transferred its Interest would be entitled, and such Assignee shall not be entitled or enabled to exercise any other rights or powers of a Member, such other rights, and all obligations relating to, or in connection with, such Interest (including the obligation to make Capital Contributions) remaining with the transferring Member. The Transferring Member shall remain a Member even if it has Transferred its entire Interest in CF LLC to one or more Assignees until such time as all such Assignees are admitted to CF LLC as

Members pursuant to Section 3.6 or Section 9.7, as applicable. In the event any Assignee desires to make a further assignment of any Interest in CF LLC, such Assignee shall be subject to all of the provisions of this Agreement to the same extent and in the same manner as the Member who initially held such Interest.

- **9.5** Admissions, Withdrawals and Removals. No Person shall be admitted to CF LLC as a Member without the prior written consent of the Manager and a Majority in Interest of the Members entitled to vote thereon, except in accordance with Section 3.6 (with respect to Persons receiving Interests directly from CF LLC) or Section 9.7 (with respect to Persons receiving Interests from a Partner or Assignee). No Person shall succeed to the rights of the Manager hereunder without the consent of the Manager and a Majority in Interest of the Members entitled to vote thereon. No Member shall be removed or entitled to withdraw from being a Member of CF LLC except in accordance with Section 3.3, this Section 9.5, Section 9.7, or Section 9.9. Except as otherwise provided in Section 10.2, no admission, withdrawal, Incapacity or removal of a Member shall cause the dissolution of CF LLC. To the maximum extent permitted by law, any purported admission, withdrawal or removal which is not in accordance with this Agreement shall be null and void.
- **9.6** <u>Admission of Assignees as Substitute Members</u>. An Assignee shall become a Substitute Member only if and when each of the following conditions is satisfied:
 - (a) The Manager consents in writing to such admission, which consent may be given or withheld, or made subject to such conditions as are determined by the Manager, all in the Manager's sole discretion; *provided* that such consent shall not be unreasonably withheld in the case of a Transfer by a Member or Assignee to an Affiliate or Immediate Family member thereof in accordance with, and subject to, Section 9.1;
 - (b) The Manager receives written instruments (including copies of any instruments of Transfer and such Assignee's consent to be bound by this Agreement as a Substitute Member) that are in form and substance satisfactory to the Manager in its sole discretion; and
 - (c) The parties to the Transfer, or any one of them, pays all of CF LLC's expenses in connection with such Transfer (including, but not limited to, the legal and accounting fees incurred by CF LLC).
- **9.7** Withdrawal of Certain Members. If a Member has Transferred all of its Interest to one or more Assignees in accordance with this Article 9, then such Member shall cease to be a member of CF LLC when all such Assignees have been admitted as Members in accordance with Section 3.6 or Section 9.6, as applicable.
- **9.8** Conversion of Interest. Upon the Incapacity of a Member, such Incapacitated Member shall automatically be converted to an Assignee only, and such Incapacitated Member (or its executor, administrator, trustee or receiver, as applicable) shall thereafter be deemed an Assignee for all purposes hereunder, with the same rights to allocations of Net Income, Net Loss and similar items and to distributions as was held by such Incapacitated Member prior to its Incapacity, but without any other rights of a Member.

9.9 <u>Limitations on Participation.</u> If the general partner of any Member determines that the partnership interest of any Fund LPs will be liquidated, redeemed, or repurchased, or that any Fund LPs shall otherwise withdraw as a partner of any Member in return for any consideration provided by such Member, in each case, pursuant to the procedures set forth in Section 9.10 of the Co-Investment Fund Agreement, the Main Fund Agreement, or the Parallel Fund Agreement, then, notwithstanding anything herein to the contrary, the Manager may cause CF LLC to take such actions as the Manager determines to be necessary or appropriate in connection therewith in order to implement such liquidation, redemption, repurchase or withdraw, including, without limitation, full or partial liquidation, redemption, or repurchase of the affected Member's Interest, payment of cash distributions to the affected Member in lieu of distributions in kind, or such other actions as the Manager determines to be necessary or appropriate.

ARTICLE 10 DISSOLUTION, LIQUIDATION AND TERMINATION OF CF LLC

- **10.1** <u>Limitations</u>. CF LLC may be dissolved, liquidated and terminated and have its affairs wound up only pursuant to the provisions of this Article 10, and the Members do hereby irrevocably waive, to the fullest extent permitted by applicable law, any and all other rights they may have to cause a dissolution of CF LLC or to maintain an action for partition of any or all of CF LLC's assets.
- **10.2** Exclusive Causes. Subject only to any non-waivable provisions of the Act, the following (and only the following) events shall cause CF LLC to be dissolved:
 - (a) The election of the Manager;
 - (b) The Incapacity of the Manager or any other event that causes the Manager to cease to be a manager under any non-waivable provision under the Act; *provided* that CF LLC shall not be dissolved or required to be wound up in connection with any of the events specified in this Section 10.2(b) if at the time of the occurrence of such event there is at least one other manager who is hereby authorized to, and elects to, carry on the business of CF LLC;
 - (c) At any time that there are no members of CF LLC, unless the business of CF LLC is continued in accordance with the Act;
 - (d) The conversion, at any time following the making of the Co-Investment, of all of the assets of CF LLC into cash;
 - (e) The dissolution of the Co-Investment Fund, the Main Fund, or the Parallel Fund, in accordance with Section 10.2 of the Co-Investment Fund Agreement, the Main Fund Agreement, or the Parallel Fund Agreement, as applicable;
 - (f) The entry of a decree of judicial dissolution under Section 18-802 of the Act; or
 - (g) Payment in full of the Preferred Distribution pursuant to Section 4.2(d).

Any dissolution of CF LLC other than as provided in this Section 10.2 shall be a dissolution in contravention of this Agreement.

- **10.3** Effect of Dissolution. The dissolution of CF LLC shall be effective on the day on which the event occurs giving rise to the dissolution, but CF LLC shall not terminate until it has been wound up, its assets have been distributed as provided in Section 10.4 and its Certificate of Formation has been canceled in accordance with the Act. Notwithstanding the dissolution of CF LLC, prior to the termination of CF LLC, the business of CF LLC and the affairs of the Members and the Manager, in their respective capacities as such, shall continue to be governed by this Agreement.
- 10.4 <u>Liquidation and Final Distribution of Proceeds</u>. Upon the dissolution of CF LLC pursuant to Section 10.2, CF LLC shall thereafter engage in no further business other than that which is necessary to wind up the business, and the Manager or, in the case of dissolution pursuant to Section 10.2(b), a liquidating trustee appointed by a Majority in Interest, after the establishment of appropriate reserves, shall liquidate all Securities and any other CF LLC assets and distribute the cash proceeds therefrom in accordance this Section 10.4. A reasonable time shall be allowed for the winding up of the affairs of CF LLC in order to minimize any losses attendant upon such a winding up. In the event the liquidator reasonably believes that it is prudent to do so, cash or other assets held in reserve may be placed in a liquidating trust or other escrow immediately prior to the termination of CF LLC in order to ensure that any and all obligations of CF LLC are satisfied. The cash proceeds from the liquidation of CF LLC assets shall be applied or distributed by CF LLC in the following order:
 - (a) First, to the creditors of CF LLC (including any Members or the Manager to the extent that they are creditors and to the extent permitted by law) of the liabilities of CF LLC other than liabilities for distributions not paid to Members;
 - (b) Second, to the Members and former Members in satisfaction (whether by payment or the making of reasonable provision for payment thereof) of liabilities, if any, for distributions not paid; and
 - (c) Third, to the Members in accordance with Section 4.1.2.

Notwithstanding the foregoing, in the event that the Manager determines that an immediate sale of all or any portion of the Securities or other CF LLC assets would cause undue loss to the Members, the Manager, in order to avoid such loss to the extent not then prohibited by the Act, may either defer liquidation of and withhold from distribution for a reasonable time any Securities or other CF LLC assets except those necessary to satisfy CF LLC's debts and obligations or, subject to the priorities set forth in Section 10.4(a) through(c) above, distribute such Securities or other CF LLC assets to the Members in kind.

10.5 <u>Post-Dissolution Restoration Obligations</u>. If any Member has a deficit balance in its Capital Account (after giving effect to all contributions, distributions and allocations for all taxable years, including the year during which the liquidation occurs), then such Member shall have no obligation to make any Capital Contribution with respect to such deficit, and such deficit shall not be considered a debt owed to CF LLC or to any other Person for any purpose whatsoever.

ARTICLE 11 MISCELLANEOUS

11.1 CF LLC Counsel. CF LLC, THE CO-INVESTMENT FUND, THE MAIN FUND, THE OTHER BDT FUNDS, THE MANAGER AND THEIR RESPECTIVE AFFILIATES MAY BE REPRESENTED BY THE SAME COUNSEL. THE ATTORNEYS, ACCOUNTANTS AND OTHER EXPERTS WHO PERFORM SERVICES FOR CF LLC MAY ALSO PERFORM SERVICES FOR THE CO-INVESTMENT FUND, THE MAIN FUND, THE PARALLEL FUND, OTHER BDT FUNDS, THE MANAGER, AND ANY AFFILIATES OF THE FOREGOING. IT IS CONTEMPLATED THAT ANY SUCH DUAL REPRESENTATION MAY CONTINUE. THE MANAGER MAY, WITHOUT THE CONSENT OF ANY MEMBER OR FUND LP, EXECUTE ON BEHALF OF CF LLC ANY CONSENT TO THE REPRESENTATION OF CF LLC, THE CO-INVESTMENT FUND, THE MAIN FUND, THE PARALLEL FUND, ANY OTHER BDT FUND, THE MANAGER, OR THEIR RESPECTIVE AFFILIATES THAT COUNSEL MAY REQUEST PURSUANT TO APPLICABLE RULES OF ETHICS AND/OR PROFESSIONAL CONDUCT OR SIMILAR RULES IN ANY APPLICABLE JURISDICTION. CF LLC HAS INITIALLY SELECTED MUNGER, TOLLES & OLSON LLP ("CF LLC COUNSEL") AS LEGAL COUNSEL TO CF LLC. EACH MEMBER ACKNOWLEDGES THAT CF LLC COUNSEL DOES NOT REPRESENT ANY MEMBER OR FUND LP IN ITS CAPACITY AS SUCH IN THE ABSENCE OF A CLEAR AND EXPLICIT WRITTEN AGREEMENT TO SUCH EFFECT BETWEEN SUCH MEMBER OR FUND LP AND CF LLC COUNSEL (AND THEN ONLY TO THE EXTENT SPECIFICALLY SET FORTH IN SUCH AGREEMENT), AND THAT IN THE ABSENCE OF ANY SUCH AGREEMENT CF LLC COUNSEL SHALL OWE NO DUTIES TO ANY MEMBER OR FUND LP. EACH MEMBER FURTHER ACKNOWLEDGES THAT. WHETHER OR NOT CF LLC COUNSEL HAS IN THE PAST REPRESENTED OR IS CURRENTLY REPRESENTING SUCH MEMBER OR ANY FUND LP WITH RESPECT TO OTHER MATTERS, CF LLC COUNSEL HAS NOT REPRESENTED THE INTERESTS OF ANY FUND LP IN THE PREPARATION AND/OR NEGOTIATION OF THIS AGREEMENT, IN CONNECTION WITH THE FORMATION OF THE CO-INVESTMENT FUND, OR OTHERWISE IN CONNECTION WITH THE CO-INVESTMENT.

11.2 Amendments.

- 11.2.1 Each additional Member shall become a signatory hereto by signing such instruments, in such manner, as the Manager shall determine. By so signing, each additional Member shall be deemed to have adopted and to have agreed to be bound by all of the provisions of this Agreement.
- 11.2.2 In addition to amendments specifically authorized herein, any provision of this Agreement may be amended by the Manager with the consent of a Majority in Interest, *provided* that: (a) any provision in this Agreement expressly requiring the consent or approval of at least 66-2/3% in Interest of the Members may be amended only by the Manager with the

consent of at least 66-2/3% in Interest; and (b) without the consent of any Member or Fund LP to be adversely affected thereby, this Agreement may not be amended so as to (i) modify the limited liability of such Member or Fund LP; (ii) disproportionately and adversely affect the interest of such Member or Fund LP in any Net Income, Net Loss or distributions; or (iii) materially, adversely and disproportionately affect the rights and entitlements of such Member or Fund LP under this Agreement.

- 11.2.3 Subject to clause (c) of Section 11.2.3, but notwithstanding any other provision in this Section 11.3, in addition to other amendments authorized herein, amendments may be made to this Agreement from time to time by the Manager, without the consent of any other Member: (a) to correct any typographical or similar ministerial errors; (b) to delete, modify or add any provision of this Agreement required to be so deleted, modified or added by, or for compliance with, applicable law or the interpretation thereof; (c) to cure any mistake or ambiguity, to correct or supplement any provision herein which may be inconsistent with any other provision herein, or to make any other provisions with respect to matters or questions arising under this Agreement that are not inconsistent with the provisions of this Agreement; (d) to take such actions as may be necessary (if any) to ensure that CF LLC will be treated as a partnership for federal income tax purposes; (e) to take such actions as may be necessary (if any) to ensure that none of the BDT Funds or the Manager will be subject to regulation under ERISA, the Investment Company Act, or the Investment Advisers Act; and (f) to give effect to the intended allocation of relative rights and obligations as between or among the BDT Funds (as contemplated by the Co-Investment Fund Agreement, the Main Fund Agreement, the Parallel Fund Agreement, and this Agreement) or as may otherwise be necessary or appropriate in a manner consistent therewith and herewith; and (g) to make any amendment hereto necessary to give effect to the admission of a Substitute Member in accordance with Section 9.6. The Manager shall provide prompt written notice of any such amendments to the Members and to the Fund LPs.
- 11.2.4 In making any amendments, there shall be prepared and filed by, or for, the Manager such documents and certificates as may be required under the Act and under the laws of any other jurisdiction applicable to CF LLC.
- 11.3 Accounting and Fiscal Year. Subject to Code Section 448, the books of CF LLC shall be kept on such method of accounting for tax and financial reporting purposes as may be determined by the Manager and consistent with Section 8.1. The fiscal year of CF LLC (the "Fiscal Year") shall be the calendar year, or, in the case of the first and last Fiscal Years of CF LLC, the fraction thereof commencing on the date hereof or ending on the date on which the winding-up of CF LLC is completed, as the case may be, unless otherwise determined by the Manager and permitted under the Code.
- **11.4** Entire Agreement. This Agreement, together with the exhibits and annexes hereto, constitutes the entire agreement between the parties hereto pertaining to the subject matter hereof and fully supersedes any and all prior or contemporaneous agreements or understandings between the parties hereto pertaining to the subject matter hereof.
- 11.5 <u>Further Assurances</u>. Each of the parties hereto hereby covenants and agrees on behalf of itself, its successors, and its assigns, without further consideration, to prepare, execute,

acknowledge, file, record, publish and deliver such other instruments, documents and statements, and to take such other action as may be required by law or reasonably necessary, to effectively carry out the purposes of this Agreement.

- 11.6 Notices. Any notice, consent, payment, demand, or communication required or permitted to be given by any provision of this Agreement shall be in writing and shall be (a) delivered personally to the Person or to an officer of the Person to whom the same is directed, (b) sent by facsimile, overnight mail or registered or certified mail, return receipt requested, postage prepaid, or (c) sent by e-mail, with electronic, written or oral confirmation of receipt, in each case addressed as follows: if to CF LLC, the Manager, or any Member c/o BDT Capital Partners, LLC, 401 North Michigan Avenue, Suite 3100, Chicago, Illinois 60611, Attn: General Counsel, facsimile no. (312) 832-1700, e-mail address, wbush@bdtcap.com, or to such other address(es) specified in writing from time to time; and if to any Fund LP, to such Fund LP in accordance with Section 11.7 of the Co-Investment Fund Agreement, the Main Fund Agreement, or the Parallel Fund Agreement, as applicable. Any such notice shall be deemed to be delivered, given and received for all purposes as of: (i) the date so delivered, if delivered personally; (ii) upon receipt, if sent by facsimile or e-mail; or (iii) on the date of receipt or refusal indicated on the return receipt, if sent by registered or certified mail, return receipt requested, postage and charges prepaid, and properly addressed.
- 11.7 <u>Governing Law</u>. This Agreement, including its existence, validity, construction, and operating effect, and the rights of each of the parties hereto, shall be governed by and construed in accordance with the laws of the State of Delaware without giving effect to any conflicts or choice of laws provisions that would cause the application of the domestic substantive laws of any other jurisdiction.
- 11.8 <u>Binding Effect</u>. Except as otherwise expressly provided herein, this Agreement shall be binding on and inure to the benefit of the parties hereto, their heirs, executors, administrators, successors and all other Persons hereafter holding, having or receiving an interest in CF LLC, whether as Assignees, Substitute Members or otherwise.
- 11.9 Anti-Money Laundering. Notwithstanding any other provision of this Agreement to the contrary, the Manager (on behalf of itself and on behalf of CF LLC) is authorized, without the consent of any Person, including any Member, to take such action as it determines in its sole discretion to be necessary or advisable to comply, or to cause CF LLC to comply, with any applicable anti-money laundering or anti-terrorist laws, rules, regulations, directives or special measures.
- **11.10** <u>Counterparts</u>. This Agreement may be executed in multiple counterparts, any of which may be delivered via facsimile, PDF, or other forms of electronic delivery, each of which shall be deemed an original and all of which, taken together, shall constitute one agreement.
- **11.11** <u>Waivers</u>. No waiver by any party hereto of any default with respect to any provision, condition or requirement hereof shall be deemed to be a waiver of any other provision, condition or requirement hereof; nor shall any delay or omission of any party to exercise any right hereunder in any manner impair the exercise of any such right accruing to it hereafter.

11.12 Severability; Preservation of Intent. If any provision of this Agreement as applied to any party or any circumstances is determined by an arbitrator or any court having jurisdiction to be void, unenforceable or inoperative as a matter of law, then the parties agree that such provision shall be modified to the greatest extent legally possible so that the intent of this Agreement may be legally carried out. If any one or more of the provisions contained herein, or the application thereof in any circumstances, is held void, unenforceable or inoperative as a matter of law in any respect or for any reason, then the validity, enforceability and operation of any such provision in every other respect and of the remaining provisions hereof shall not be in any way impaired or affected, it being intended that all of the parties' rights and privileges shall be enforceable to the fullest extent permitted by law. Notwithstanding the foregoing, if: (a) one or more of the provisions contained herein, or the application thereof to any party or in any circumstances, is held void, unenforceable or inoperative as a matter of law in any respect or for any reason, and (b) as a result, the Manager, in its reasonable discretion, determines that the basic intent and purpose of this Agreement is irreparably frustrated thereby, then the Manager may cause CF LLC to dissolve pursuant to Section 10.2(a).

11.13 Certain Rules of Construction. To the fullest extent permitted by law, the parties hereto intend that any ambiguities shall be resolved without reference to which party may have drafted this Agreement. All Article or Section titles or other captions in this Agreement are for convenience only, and they shall not be deemed part of this Agreement and in no way define, limit, extend or describe the scope or intent of any provisions hereof. Unless the context otherwise requires: (a) a term has the meaning assigned to it; (b) an accounting term not otherwise defined has the meaning assigned to it in accordance with thenapplicable United States generally accepted accounting principles; (c) "or" is not exclusive; (d) words in the singular include the plural, and words in the plural include the singular; (e) provisions apply to successive events and transactions; (f) the words "herein," "hereof" and other words of similar import refer to this Agreement as a whole and not to any particular Article, Section or other subdivision; (g) all references herein to Articles, Sections, Exhibits, Annexes, paragraphs, subparagraphs and clauses shall be deemed to be references to Articles, Sections, paragraph, subparagraphs and clauses of, and Exhibits and Annexes to, this Agreement unless the context shall otherwise require; (h) any pronoun used in this Agreement shall include the corresponding masculine, feminine or neuter forms; (i) the words "include," "includes" and "including" shall be deemed to be followed by the phrase "without limitation"; (j) the word "extent" in the phrase "to the extent" shall mean the degree to which a subject or other thing extends, and such phrase shall not mean simply "if"; (k) references to "\$" or "dollars" shall mean United States dollars; (1) unless otherwise expressly provided herein, any agreement, instrument or statute defined or referred to herein or in any agreement or instrument that is referred to herein means such agreement, instrument or statute as from time to time amended, modified or supplemented, including (in the case of agreements or instruments) by waiver or consent and (in the case of statutes) by succession of comparable successor statutes and references to all attachments thereto and instruments incorporated therein; and (m) all references to any Member shall mean and include such Member and any Person duly admitted as a member in CF LLC in substitution therefor in accordance with this Agreement, unless the context otherwise requires.

11.14 <u>Jurisdiction</u>. Each party hereby submits to the nonexclusive jurisdiction of any state or federal court sitting in the State of Delaware in any action or proceeding arising out of or relating to this Agreement or the transactions contemplated herein. To the maximum extent

permitted by applicable law, any final judgment against a party in any proceedings brought in the United States will be conclusive and binding upon such party and may be enforced against such party in the courts of any other jurisdiction. Nothing in this Section 11.14 limits the rights of CF LLC or any party to commence any proceedings or to serve process by another manner permitted by law in any other court of competent jurisdiction; nor will the bringing or continuing of proceedings in one or more jurisdictions preclude the bringing or continuing of proceedings in any other jurisdiction, whether concurrently or otherwise. Each party's obligation under this Section 11.14 will survive the dissolution, liquidation and winding up of CF LLC.

11.15 <u>Survival</u>. The provisions of Sections 3.5, 6.5, 8.3, 11.1, 11.5, 11.6, 11.7, and 11.14, and this Section 11.15 (and any other provisions herein necessary for the effectiveness of the foregoing sections) shall survive the termination of CF LLC and/or the termination of this Agreement.

[Signature Pages Follow]

IN WITNESS WHEREOF, the parties hereto have duly executed this Agreement as of the day and year first above written.

MANAGER:

BDTCP GP I, LLC

By: BDT Capital Partners, LLC, its managing member

By: /s/ William R. Bush

Name: William R. Bush

Title: Vice President, General Counsel and Secretary

MEMBERS:

BDT CAPITAL PARTNERS FUND I, L.P.

By: BDTCP GP I, LLC, its general partner

By: BDT Capital Partners, LLC, its managing member

By: /s/ William R. Bush

Name: William R. Bush

Title: Vice President, General Counsel and Secretary

BDT CAPITAL PARTNERS FUND I-A, L.P.

By: BDTCP GP I, LLC, its general partner

By: BDT Capital Partners, LLC, its managing member

By: /s/ William R. Bush

Name: William R. Bush

Title: Vice President, General Counsel and Secretary

BDT CAPITAL PARTNERS CF FUND I, L.P.

By: BDTCP GP I, LLC, its general partner

By: BDT Capital Partners, LLC, its managing member

By: /s/ William R. Bush

Name: William R. Bush

Title: Vice President, General Counsel and Secretary

ANNEX I

ADDITIONAL ALLOCATION PROVISIONS

- **1.1** Additional Allocation Provisions. Notwithstanding any provisions of Article 5:
- 1.1.1 Regulatory Allocations.
- (a) *Minimum Gain Chargeback*. Except as otherwise provided in Reg. §§ 1.704-2(f), notwithstanding the provisions of Section 5.2, or any other provision of Article 5 or this Annex I, if there is a net decrease in partnership minimum gain (as defined in Reg. § 1.704-2(b)(2) and as determined under Reg. § 1.704-2(d)) during any Fiscal Year, each Member shall be specially allocated items of CF LLC income and gain for such year (and, if necessary, subsequent years) in an amount equal to such Member's share of the net decrease in partnership minimum gain, as determined under Reg. § 1.704-2(g). Allocations pursuant to the previous sentence shall be made in proportion to the respective amounts required to be allocated to each Member pursuant thereto. The items to be allocated shall be determined in accordance with Reg. §§ 1.704-2(f)(6) and 1.704-2(j)(2). This Section 1.1.1(a) of this Annex I is intended to meet the minimum gain chargeback requirements of Reg. § 1.704-2(f) and shall be interpreted consistently therewith.
- (b) Partner Minimum Gain Chargeback. Except as otherwise provided in Reg. § 1.704-2(i)(4), and notwithstanding any other provision of Article 5 or this Annex I (except Section 1.1.1(a) of this Annex I), if there is a net decrease in partner minimum gain (determined under Reg. § 1.704-2(i)(2)) attributable to a partner nonrecourse debt (as defined in Reg. § 1.704-2(b)(4)) during any Fiscal Year, each Member who has a share of such partner minimum gain, determined under Reg. § 1.704-2(i)(5), shall be specially allocated items of CF LLC income and gain for such year (and, if necessary, subsequent years) in an amount equal to such Member's share of the net decrease in such partner minimum gain, determined under Reg. § 1.704-2(i)(4). Allocations pursuant to the previous sentence shall be made in proportion to the respective amounts required to be allocated to each Member pursuant thereto. The items to be so allocated shall be determined in accordance with Reg. §§ 1.704-2(i)(4) and 1.704-2(j)(2). This Section 1.1.1(b) of this Annex I is intended to meet the partner nonrecourse debt minimum gain chargeback requirement of Reg. § 1.704-2(i)(4) and shall be interpreted consistently therewith.
- (c) *Nonrecourse Deductions and Partner Nonrecourse Deductions*. Any nonrecourse deductions (as defined in Reg. §§ 1.704-2(b)(1) and 1.704-2(c)) for any Fiscal Year shall be specially allocated to the Members in accordance with their Percentage Interests. Any partner nonrecourse deductions (as defined in Reg. § 1.704-2(i)) for any Fiscal Year shall be specially allocated to the Member(s) who bear(s) the economic risk of loss (within the meaning of Reg. § 1.752-2) with respect to the partner nonrecourse debt to which such partner nonrecourse deductions are attributable, in accordance with Reg. § 1.704-2(i).
- (d) *Qualified Income Offset*. If any Member unexpectedly receives an adjustment, allocation or distribution described in Reg. §§ 1.704-1(b)(2)(ii)(d)(4), (5) or (6), items of CF LLC income and gain shall be allocated, in accordance with Reg. § 1.704-1(b)(2)(ii)(d), to the Member in an amount and manner sufficient to eliminate, to the extent required by such Section,

the deficit balance in the Adjusted Capital Account of the Member as quickly as possible, *provided* that an allocation pursuant to this Section 1.1.1(d) of this Annex I shall be made if and only to the extent that such Member would have a deficit balance in its Adjusted Capital Account after all other allocations provided in Article 5 have been tentatively made as if this Section 1.1.1(d) of this Annex I were not in this Agreement. It is intended that this Section 1.1.1(d) of this Annex I qualify and be construed as a "qualified income offset" under Reg. § 1.704-1(b)(2)(ii)(d) and shall be interpreted consistently therewith.

- (e) Capital Account Deficits. Notwithstanding Article 5 or the foregoing provisions of this Annex I, a Member shall not be allocated its portion of any item of Net Loss to the extent that such allocation would cause or increase a deficit balance in its Adjusted Capital Account. Any item of Net Loss or portion thereof which, but for the preceding, would be allocated to a Member, shall be allocated to each Member having a positive balance in its Adjusted Capital Account, to the extent of such positive balance, in proportion to such positive Capital Account balances.
- (f) Section 754 Adjustment. To the extent an adjustment to the adjusted tax basis of any CF LLC asset pursuant to Code Section 734(b) or Code Section 743(b) is required, pursuant to Reg. §§ 1.704-1(b)(2)(iv)(m)(2) or 1.704-1(b)(2)(iv)(m)(4), to be taken into account in determining Capital Accounts as the result of a distribution to a Member in complete liquidation of its Interest in CF LLC, the amount of such adjustment shall be treated as an item of gain (if the adjustment increases the basis of the asset) or loss (if the adjustment decreases such basis) and such gain or loss shall be specially allocated to the Members in accordance with their Interests in CF LLC, if Reg. § 1.704-1(b)(2)(iv)(m)(2) applies, or to the Members to whom such distribution was made, if Reg. § 1.704-1(b)(2)(iv)(m)(4) applies.
- (g) Curative Allocation. The allocations set forth in this Section 1.1.1 of this Annex I are intended to comply with certain regulatory requirements, including Reg. §§ 1.704-1(b) and 1.704-2 and shall be interpreted consistently therewith. Notwithstanding the provisions of Section 5.2, such allocations shall be taken into account in allocating other items of income, gain, loss and deduction among the Members so that, to the extent possible, the net amount of such allocations of other items and these allocations of this Section 1.1 of this Annex I to each Member shall be equal to the net amount that would have been allocated to each such Member if such allocations had not occurred.
- 1.1.2 Excess Nonrecourse Liabilities. For purposes of determining a Member's proportional share of the "excess nonrecourse liabilities" of CF LLC within the meaning of Reg. § 1.752-3(a)(3), each Member's interest in CF LLC profits shall be such Member's Percentage Interest.

1.2 Tax Allocations.

1.2.1 *In General*. Except as otherwise provided in this Section 1.2 of this Annex I, each item of income, gain, loss, deduction and credit shall be allocated among the Members for income tax purposes in the same manner as its correlative item of "book" income, gain, loss, deduction or credit is allocated pursuant to Section 5.2 and Section 1.1 of this Annex I.

1.2.2 Allocations Respecting Section 704(c) Revaluations. Notwithstanding Section 1.2.1 of this Annex I, items of income, gain, loss, deduction and credit with respect to CF LLC property that is contributed to CF LLC by a Member shall be shared among the Members for income tax purposes in accordance with Code Section 704(c) and the regulations thereunder, so as to take into account, under any permitted method chosen by the Manager, any variation between the basis of the property to CF LLC and its initial Gross Asset Value. In the event the Gross Asset Value of any CF LLC asset is adjusted pursuant to subparagraph (b) of the definition of Gross Asset Value, subsequent allocations of items of income, gain, loss and deduction with respect to such asset for income tax purposes shall take account of any variation between the adjusted basis of such asset and its Gross Asset Value in accordance with Code Section 704(c) and the applicable regulations thereunder, under any permitted method chosen by the Manager.

1.3 Other Provisions.

- 1.3.1 For any Fiscal Year during which any part of a Interest is transferred between the Members or to another Person, the portion of the Net Income, Net Loss and other items of income, gain, loss, deduction and credit that are allocable with respect to such part of a Interest shall be apportioned between the transferor and the transferee under any method allowed pursuant to Code Section 706 and the regulations thereunder as determined by the Manager.
- 1.3.2 The provisions of this Agreement relating to the maintenance of Capital Accounts are intended to comply with Code Section 704 and the regulations thereunder, and will be interpreted and applied in a manner consistent with such regulations. In the event that the Manager in good faith determines that the Code or any such regulation requires allocations of items of income, gain, loss, deduction or credit different from those set forth in Article 5 or this Annex I, the Manager is hereby authorized to make new allocations in reliance on the Code and such regulations, and no such new allocation shall give rise to any claim or cause of action by any Member against the Manager or CF LLC *provided*, *however*, if any new allocation will cause a material, disproportionate adverse effect on any Member or Fund LP, the Manager will consult with such Member or Fund LP prior to making any adjustment or modification.
- 1.3.3 The Members acknowledge and are aware of the income tax consequences of the allocations made by Article 5 or this Annex I and hereby agree to be bound by the provisions of Article 5 or this Annex I in reporting their shares of Net Income, Net Loss and other items of income, gain, loss, deduction and credit for federal, state and local income tax purposes.

[End of Annex I]

JOINT FILING AGREEMENT

January 23, 2012

In accordance with Rule 13d-1(k)(1) promulgated under the Securities Exchange Act of 1934, as amended, the undersigned hereby agree to the joint filing with all other Reporting Persons (as such term is defined in the Schedule 13D referred to below) on behalf of each of them of a statement on Schedule 13D (including amendments thereto) with respect to the Common Stock, par value \$0.001 per share, of Colfax Corporation, a Delaware corporation, and that this agreement may be included as an exhibit to such joint filing.

IN WITNESS WHEREOF, the undersigned hereby execute this Agreement as of January 23, 2012.

BDT CAPITAL PARTNERS, LLC

By: /s/ Byron D. Trott

Name: Byron D. Trott

Its: President and Chief Investment Officer

BDTCP GP I, LLC

By: /s/ Byron D. Trott

Name: Byron D. Trott

Its: President and Chief Investment Officer

BDT CF ACQUISITION VEHICLE, LLC

By: BDTCP GP I, LLC

Its: Manager

By: /s/ Byron D. Trott

Name: Byron D. Trott

Its: President and Chief Investment Officer

BYRON D. TROTT

/s/ Byron D. Trott

BDTP GP, LLC

By: /s/ Byron D. Trott

Name: Byron D. Trott

Its: President and Chief Investment Officer